

UNIVERSIDAD DE CHILE FACULTAD DE CIENCIAS FÍSICAS Y MATEMÁTICAS DEPARTAMENTO DE INGENIERÍA ELÉCTRICA

# DESIGNING AND MODELING OF A COEXISTENCE MECHANISM FOR VEHICULAR COMMUNICATIONS ON TV WHITE SPACES

### TESIS PARA OPTAR AL GRADO DE DOCTOR EN INGENIERÍA ELÉCTRICA

## ADRIANA ARTEAGA ARCE

### PROFESORA GUÍA: DRA. SANDRA LORENA CÉSPEDES UMAÑA

### PROFESOR CO-GUÍA: DR. CÉSAR AUGUSTO AZURDIA MEZA

### MIEMBROS DE LA COMISIÓN: DR. CLAUDIO IGNACIO ESTÉVEZ MONTERO DRA. NATHALIE MITTON DR. DIEGO DUJOVNE HELMAN

#### SANTIAGO DE CHILE 2021

#### SUMMARY OF THE THESIS SUBMITTED FOR THE DEGREE OF DOCTOR IN ELECTRICAL ENGINEERING BY: ADRIANA ARTEAGA ARCE DATE: 2021 ADVISOR: PROF. DRA. SANDRA LORENA CÉSPEDES UMAÑA CO-ADVISOR: PROF. DR. CÉSAR AUGUSTO AZURDIA MEZA

# DESIGNING AND MODELING OF A COEXISTENCE MECHANISM FOR VEHICULAR COMMUNICATIONS ON TV WHITE SPACES

The expected growth in the number of connected vehicles and the data generated by applications and information systems will require alternative technologies to expand the spectrum for communications. The IEEE 802.22 is a standard for wireless regional area networks to operate in TV bands opportunistically; however, sharing the TV White Space (TVWS) spectrum creates a problem of coexistence among secondary networks because no mechanism is defined for coordinating channel usage. This thesis addresses the challenge of enabling the coexistence of fixed secondary and vehicular networks when no vacated TVWS channels are available for allocation, and both networks share the same TVWS channel. The proposed White Space Resource Sharing mechanism (WSRS) based on power control and channel access allocates resources to vehicles even when the density of fixed nodes and the transmission power of the fixed network cause high interference to the vehicular network. The WSRS mechanism implements a strategy to protect the fixed secondary network from interference caused by vehicular transmissions while promoting that both networks simultaneously use the channel. The offered channel capacity to the vehicular network increases up to 40%using the WSRS mechanism compared to channel capacity when TVWS channel sharing is not allowed.

RESUMEN DE LA TESIS PARA OPTAR AL GRADO DE DOCTOR EN INGENIERÍA ELÉCTRICA POR: ADRIANA ARTEAGA ARCE FECHA: 2021 PROFESORA GUÍA: DRA. SANDRA LORENA CÉSPEDES UMAÑA PROFESOR CO-GUÍA: DR. CÉSAR AUGUSTO AZURDIA MEZA

#### DISEÑO E IMPLEMENTACIÓN DE UN MECANISMO DE COEXISTENCIA PARA COMUNICACIONES VEHICULARES EN ESPACIOS EN BLANCO DE TELEVISIÓN

El crecimiento esperado en el número de vehículos conectados y los datos generados por las aplicaciones y los sistemas de información requerirán tecnologías alternativas para expandir el espectro de comunicaciones. El estándar IEEE 802.22 especifica la operación oportunista de redes inalámbricas de área regional (WRAN) en las bandas de televisión; sin embargo, compartir el espectro de espacios en blanco de televisión (TVWS por sus siglas en inglés) crea el problema de coexistencia entre redes secundarias porque el mecanismo para coordinar el uso del canal TVWS no está definido. Esta tesis aborda el desafío de habilitar la coexistencia entre redes fijas y redes vehiculares secundarias cuando no hay canales de TVWS disponibles para asignar, y ambas redes deben compartir el canal TVWS. Se propone el mecanismo White Space Resource Sharing mechanism (WSRS) basado en control de potencia y acceso al canal, que asigna recursos a los vehículos aún cuando la densidad de los nodos fijos y su potencia de transmisión ocasionan alta interferencia a la red vehicular. El mecanismo WSRS implementa una estrategia para proteger a la red fija secundaria de la interferencia causada por la red vehicular, mientras promueve que ambas redes continúen utilizando el canal TVWS. La capacidad ofrecida a la red vehicular incrementa en hasta un 40% usando el mecanismo WSRS comparada con la capacidad de canal cuando el uso compartido no está habilitado.

ii

Dedicada a la Luz Suprema del Amor Supremo. A mis amados papá, mamá, hermana y hermano. A mi. Ananda.

#### AGRADECIMIENTOS

Infinita Gratitud y más allá de las palabras a la Divinidad por guiarme con Su Mano amorosa, paciente y llena de Gracia en este camino. Infinita Gratitud a la Energía Suprema que fluye a través de mi. Sigo siendo un instrumento para expandir tu Existencia, Conciencia y Dicha.

Infinita Gratitud a mi supervisora, Sandra Lorena Céspedes Umaña, por su apoyo sin condiciones, guía, compañía e inspiración. Gracias por abrir tantas puertas a lo largo del camino, por su paciencia, comprensión y real esfuerzo para entender lo que quería hacer, aclarar el camino, y contribuir con sus sugerencias, experiencia y recursos. Gracias por brindarme el tiempo y el espacio para retomar el impulso, continuar y terminar. Ha sido un honor y una alegría seguirnos cruzando en el camino.

Infinita Gratitud a mi co-supervisor, César Augusto Azurdia Meza, por su apoyo, paciencia, contribución, comprensión y real interés para participar en esta investigación. Por su rigurosidad en el desarrollo investigativo, su interés genuino por el futuro de sus estudiantes y porque siempre pone a disposición sus capacidades y experiencia.

Infinita Gratitud a Papá y Mamá por ser la fuente de tanto Amor y Alegría, gracias por su apoyo sin condiciones, porque su Amor me acompaña siempre y han sido una inspiración para mi. Es un honor y una bendición ser su hija. Infinita Gratitud a mi hermana y hermano porque sus vidas me inspiran, por acompañarme y cuidarme cuando los he necesitado. Infinita Gratitud a mis ancestros por sus dones, protección y liberación. Que la Luz sea su hogar.

Infinita Gratitud a quienes me han acompañado a mi paso por Chile y por todos los lugares a los que se me ha ocurrido ir. Especialmente a Jeanina, Patricia y Oscar, porque la Conciencia nos reencontró nuevamente y nos une en su Intención. Amor, Gratitud y Unidad siempre.

Infinita Gratitud a mis amigos y compañeros de la Universidad de Chile, especialmente a Tomás y Pablo por contribuir en publicaciones conjuntas. Muchas gracias a todos por su compañía y amistad. Que todos alcancen su máximo potencial en sus caminos.

Infinita Gratitud a Javier por su apoyo, amistad y por ser el manager de mi carrera artística. Infinita Gratitud a Patricia por todo el apoyo administrativo y su cariño. Infinita Gratitud a Antonina, que con su apoyo, guía y confianza me acompañó a cerrar este ciclo.

Infinita Gratitud a CONICYT/ANID por la financiación otorgada para realizar mis estudios de doctorado con el Programa de Beca Doctorado Nacional No. 2016-21161383. Extiendo los agradecimientos al Centro Avanzado en Ingeniería Eléctrica y Electrónica, AC3E, Proyecto Basal FB0008, ANID por el apoyo en la realización de este trabajo.

Infinita Gratitud a la Energía Suprema que viene a través de esta tesis. Gracias por traerme tantas enseñanzas a tu paso, por quedarte conmigo y confiar. Que todo lo que venga de ti sea luminoso, abundante, y que las manos y conciencias que te sigan trabajando te entiendan y sostengan.

Con Amor y Gratitud, Adriana Arteaga Arce.

# Contents

1	Intr	roduction	1						
	1.1	Motivation	2						
	1.2	Problem definition	3						
	1.3	Hypotheses	3						
	1.4	Objectives	4						
		1.4.1 General Objective	4						
		1.4.2 Specific Objectives	4						
	1.5	Contributions	4						
	1.6	Publications	5						
	1.7	Outline of the thesis	5						
<b>2</b>	Lite	erature review	7						
	2.1	Vehicular Networks	7						
		2.1.1 Applications and services	8						
		2.1.2 Communication technologies	9						
	2.2	Cognitive Vehicular Networks	13						
	2.3	Dynamic Spectrum Access for Vehicular Networking	4						
		2.3.1 Available spectrum estimation mechanisms	16						
		2.3.2 Spectrum estimation based on the infrastructure	19						
		2.3.3 Spectrum estimation based on the vehicles	20						
		2.3.4 Resource allocation mechanisms	21						
	2.4	IEEE standard 802.22							
	2.5	Secondary opportunistic networks over TVWS and coexistence	25						
	2.6	Discussion	26						
3	Coe	existence of Cognitive Networks for Vehicular Communications on TV							
	$\mathbf{W}\mathbf{h}$	ite Spaces 2	28						
	3.1	Introduction	28						
	3.2	Modeling primary and secondary TV bands users for vehicular networking on							
		TVWS	29						
		3.2.1 TV primary users	30 20						
		3.2.2 White-Fi network description and traffic modeling	32						
		3.2.3 Vehicular network modeling	35						
	3.3	Coexistence between fixed and mobile (vehicular) secondary groups on the							
		IEEE standard 802.22	37						

	3.4	White	e Space Resource Sharing (WSRS) Mechanism	39
		3.4.1	System's operation	39
		3.4.2	Resource allocation in WSRS	40
4	Eva	luatio	n and Results	44
	4.1	Feasib	ility study of TVWS spectrum sharing for vehicular communications on	
		TVW	S spectrum	44
		4.1.1	Evaluation scenario	44
		4.1.2	Simulation Results	46
		4.1.3	Discussion	50
	4.2	Evalu	ation of Opportunistic Access Strategies for Vehicular Networking Over	
		TVW	S spectrum	52
		4.2.1	Scenario of Study	53
		4.2.2	Simulation results	55
		4.2.3	Discussion	57
	4.3	Perfor	mance Evaluation of the White Space Resource Sharing mechanism	57
		4.3.1	Evaluation scenario	57
		4.3.2	Simulation Results	60
		4.3.3	Discussion	67
<b>5</b>	Cor	nclusio	ns	70
	5.1	Major	Research Results	70
	5.2	Futur	e Work	72
6	App	pendix		74
Bi	ibliog	graphy		75

# List of Tables

2.1	Quality of service requirements of vehicular applications and services. Adapted				
	from [26]	9			
2.2	10 MHz OFDM channel parameters for IEEE 802.11p [6]	11			
2.3	Data rate options in a DSRC 10 MHz OFDM channel [6]	11			
2.4	IEEE 802.22 PHY and MAC Parameters [23]	23			
2.5	Comparison parameters of 802.11af and 802.22 standards	25			
3.1	Summary of variables in Hata path loss model and $K_{Davidson}$ derivation	31			
3.2	WSRS mechanism symbols and variables	41			
4.1	Evaluation Parameters	46			
$4.1 \\ 4.2$	Evaluation Parameters	$46 \\ 47$			
4.1 4.2 4.3	Evaluation Parameters	$46 \\ 47 \\ 49$			
<ol> <li>4.1</li> <li>4.2</li> <li>4.3</li> <li>4.4</li> </ol>	Evaluation Parameters	46 47 49			
$ \begin{array}{r} 4.1 \\ 4.2 \\ 4.3 \\ 4.4 \end{array} $	Evaluation Parameters	46 47 49 50			
$ \begin{array}{c} 4.1 \\ 4.2 \\ 4.3 \\ 4.4 \\ 4.5 \end{array} $	Evaluation Parameters	46 47 49 50			
<ul> <li>4.1</li> <li>4.2</li> <li>4.3</li> <li>4.4</li> <li>4.5</li> </ul>	Evaluation Parameters	46 47 49 50 50			
<ul> <li>4.1</li> <li>4.2</li> <li>4.3</li> <li>4.4</li> <li>4.5</li> <li>4.6</li> </ul>	Evaluation Parameters	46 47 49 50 50 50			

# List of Figures

2.1	DSRC frequency band allocation. Source: Own elaboration based on [6].	10
2.2	General architecture for vehicular dynamic spectrum access. Source: Own elaboration	15
2.3	Advanced Wireless Rural Area Network (A-WRAN). Source: Own elaboration         based on [23]	24
3.1	Scenario of study: opportunistic access for vehicular networking considering the presence of primary TV users and secondary White-Fi networks over the TV band Source: Own elaboration	30
3.2	Received power depending on the distance between a TV transmitter and a potential TV receiver.	33
3.3	Received power depending on the distance between the White-Fi access point and a potential TVWS receiver.	34
3.4	Examples of channel occupation policies: $(a, b)$ uniformly distributed proba- bility of busy channel and $c$ ) burst traffic	35
3.5	Differences in channel verification times for two vehicles traveling at 30 km/h and 90 km/h to cover a distance of 400 m. The channel occupation policy	
3.6	follows a Pareto Distribution. Source: Own elaboration	36
	single A-BS. Source: Own elaboration.	37
3.7	Scenario of study. Source: Own elaboration.	38
$3.8 \\ 3.9$	White Space Resource Sharing (WSRS) Mechanism. Source: Own elaboration. Upstream sub-frame scheduling for vehicular and fixed network. Source: Own	39
	elaboration.	40
3.10	Power control function variables. Source: Own elaboration	42
4.1	Evaluation scenario. Source: Own elaboration	45
4.2	CAFOVA in uniformly distributed channel occupations $P_{bc}$ , when v is set to 20 km/h 60 km/h 90 km/h and 120 km/h $l = 100$ m	48
4.3	CAFOVA in Pareto distributed channel occupations A, B, and C, when the	10
4.4	area type is urban, suburban, and rural. $v = 50 \text{ km/h}$ , $l = 100 \text{ m}$ CAFOVA in uniformly distributed channel occupations $P_{bc}$ , when $l$ is set to	49
	25 m, 50 m, 100m, and 200 m. $v = 50$ km/h	51
4.5	CAFOVA in Pareto distributed channel occupations A, B, and C, when l is set to 25 m 50 m 100m and 200 m $v = 50$ km/h	52
4.6	Scenario of study. Source: Own elaboration.	53
	•	

4.7	Channel access delay for strategy A and strategy B depending on the proba-	
	bility of busy channel $(P_B)$	56
4.8	Packet loss rate for strategy A and strategy B depending on the probability	
	of busy channel $(P_b)$	57
4.9	Simulated scenario. Source: Own elaboration.	58
4.10	Fixed node distribution within the network coverage area. Source: Own elab-	
	oration.	59
4.11	Normalized Average Channel Capacity per time frame for the set of vehicles	61
4.12	Service rate for the set of vehicles	63
4.13	Average Jain's Index per time frame for the set of vehicles	65
4.14	Spectral efficiency for the FNS and VN system	66

# List of Abbreviations

3GPP	GPP Third Generation Partnership Project			
$5\mathrm{G}$	Fifth Generation			
A-BS Advanced Base Station				
A-CPE	Advanced Consumer Premise Equipment			
A-WRAN	Advanced Wireless Regional Access Network			
ADPS Adaptive Data Pipe Selection				
AP Access Point				
AWGN	Additive White Gaussian Noise			
BS	Base Station			
BPSK	Binary Phase Shift Keying			
C-V2X	Cellular Vehicle-to-Everything			
CAFOVA	Channel Availability for Opportunistic Vehicular Access			
$\mathbf{CAP}$	Channel Allocation Process			
CCH Control Channel				
CMRS Commercial Mobile Radio Service				
CEPT	Conférence Européenne des Administrations des Postes et des			
	Télécommunications			
$\mathbf{CPE}$	Consumer Premise Equipment			
CSMA/CA	Carrier Sense Multiple Access with Collision Avoidance			
$\mathbf{CSN}$	Cognitive Secondary Network			
CVN	Cognitive Vehicular Network			
D2D	Device-to-Device			
DAMA	Demand Assigned Multiple AccesS			
DSA	Dynamic Spectrum Access			
DSRC	Dedicated short-range communications			
FCC	Federal Communications Commission			
$\mathbf{FSN}$	Fixed Secondary Network			
$\operatorname{GL}$	Geolocation Function			
GDBS	Geolocation Database Server			

$\operatorname{HetVNETs}$	Heterogeneous Vehicular Ad Hoc Networks			
IEEE	Institute of Electrical and Electronics Engineers			
IIoT	industrial Internet of Things			
ITS	Intelligent Transport Systems			
LIDAR	Laser Imaging Detection and Ranging			
LiFi	Light-Fidelity			
LLC	Logical Link Control			
LOS	Line-of-sight			
LTE	Long Term Evolution			
MAC	Medium Access Control			
mmwave	Millimeter Waves			
MNO	Mobile Network Operator			
$\mathbf{MV}$	Manager Vehicle			
NLOS	Non-line-of-sight			
OBU	On-board Unit			
OFDM	Orthogonal Frequency Division Multiplexing			
OFDMA	Orthogonal Frequency Division Multiple Access			
PCP	Power Control Process			
PLCP	Physical Layer Convergence Procedure			
PLMS	Private Land Mobile Service			
PMD	Physical medium dependence			
PHY	Physical Layer			
PHY-OM1	Physical Layer Operation Mode 1			
PHY-OM2	Physical Layer Operation Mode 2			
$\mathbf{QAM}$	Quadrature Amplitude Modulation			
QPSK	Quadrature Phase Shift Keying			
$\mathbf{QoS}$	Quality of Service			
$\mathbf{RAF}$	Resource Allocation Function			
R-CPE	Relay Consumer Premise Equipment			
RLSS	Registered Location Secure Server			
$\mathbf{RSU}$	Road Site Unit			
$\mathbf{SM}$	Spectrum Manager			
$\mathbf{SCH}$	Service Channel			
$\mathbf{SDN}$	Software-Defined Network			
$\mathbf{SDR}$	Software-Defined Radio			
SINR	Signal-to-Interference Plus Noise Ratio			
SIUF	Scheduling Information Updating Function			
$\mathbf{SNR}$	Signal-to-Noise Ratio			
$\mathbf{SSF}$	Spectrum Sensing Function			

SULF	Spectrum Usage Learning Function		
$\operatorname{TDD}$	Time Division Duplex		
TDMA	Time Division Multiple Access		
$\mathbf{TVWS}$	TV White Spaces		
TVWS CH	TV White Space Channel		
UE	User Equipment		
UHF	Ultra High Frequency		
V2I	Vehicle-to-Infrastructure		
V2V	Vehicle-to-Vehicle		
V2X	Vehicle-to-Everything		
VANETs	Vehicular Ad Hoc Networks		
VCN	Vehicular Communications Network		
VDSA	Vehicular Dynamic Spectrum Access		
$\mathbf{VHF}$	Very High Frequency		
$\mathbf{VN}$	Vehicular Network		
WLAN	Wireless Local Area Network		
$\mathbf{White} extsf{-}\mathbf{Fi}$	Wifi over TV White spaces		
WHO	World Health Organization		
WRAN	Wireless Regional Access Network		
WSRS	White Space Resource Sharing		

# Chapter 1

# Introduction

Vehicular communications are considered one of the foundations to support the deployment of Intelligent Transportation Systems (ITS). Transportation systems are crucial for economic productivity and sustainable urban and rural development since these systems move food, raw materials, supplies, and assets as well as people in public and private transportation. Innovative applications for vehicular environments promise to help increase safety and comfort for passenger and drivers, to the point that government offices, such as the U.S. National Highway Traffic Safety Administration, are recommending the mandatory use of vehicular communications shortly [1]. Vehicular communications allow the exchange of information in transportation systems, improving safety and mobility on the route through coordination among vehicles and providing information and entertainment services for drivers and passengers.

According to the market research report from Markets and Markets released in 2020, the global connected car market size is expected to be USD 99 billion in 2022 and is projected to reach USD 166.0 billion by 2025 [2]. BI Intelligence in March 2020 reports that over 77 million connected cars will be on the road by 2025 [3]. The vehicles are connected through a built-in antenna and chipset or using hardware to allow drivers to connect to their cars via a smartphone. Also, connected vehicles have several systems and sensors that generate a large amount of data such as the pedal travel sensor, the vision system, the proximity sensors, among others. In the case of autonomous vehicles, each vehicle will be generating approximately 4,000 GB of data a day due to the cameras, radar, sonar, GPS, and laser imaging detection and ranging (LIDAR) components [4]. All these data is used to keep the vehicle's performance and safety, and vehicles share the data with other vehicles and with the road infrastructure to support vehicular applications, providing better infotainment services and enhancing road safety and efficiency. All efforts will improve road safety for drivers, pedestrian, cyclists, and motorcycles, promoting the coexistence between vehicles and people.

There are several challenges for achieving the ITS deployment. One of them is to enable the communication between vehicles, the signalizing infrastructure, and the other information systems on the road. Vehicular communication networks (VCN) correspond to networks formed by vehicles where communications are vehicle-to-vehicle (V2V) or vehicleto-infrastructure (V2I) [5]. Several technologies have been explored to allow VCN through cellular or WiFi networks, meanwhile standardization bodies have developed the dedicated short-range communication (DSRC) technology for V2V and V2I communications [6] and cellular vehicle-to-everything (C-V2X) [7]. In 2020, the Federal Communications Commission (FCC) decreased the spectrum allocation for vehicular communications in the 5.9 GHz band from 75 MHz to 30 MHz (5,890-5,925 GHz) for enhanced automobile safety using C-V2X technology instead of DSRC [8]. Modification addresses the automotive market to adopt C-V2X for vehicular communications, leaving aside DSRC. However, there are still 70 MHz allocated to DSRC technology in Europe [9] and 80 MHz in Japan [10] in the 5.8 GHz frequency band. Nevertheless, the expected growth in the number of connected vehicles and the data generated by applications and information systems have led to exploring alternative technologies to expand the spectrum for communications. The scarcity of radio spectrum is a problem for several services based on wireless technologies [11], including vehicular communications, making the secondary use of spectrum on television bands, known as TV White Spaces (TVWS), a key alternative to extend the available bandwidth [12]. Secondary spectrum access uses available licensed frequencies when the spectrum is underutilized or when the licensed users are inactive in the area [13].

## 1.1 Motivation

The benefits of exploiting connectivity in TV bands have brought attention to TVWS technologies and the deployment of cognitive secondary networks over the TVWS spectrum [14]. Recent studies have proposed the deployment of secondary networks to improve broadband access in rural areas [15, 16], to provide connectivity for educational initiatives [17, 18, 19] and emergency response plans [20], and to support agricultural projects [21]. Furthermore, commercial initiatives and manufacturers such as a5Systems, 6HARMONICS, Airband, and First Broadband provide technological services for rural broadband connectivity, the Industrial Internet of Things (IIoT), communications for critical infrastructure and environmental monitoring, among other applications, all operating over TVWS. However, the increasing deployment of such secondary networks creates a problem of coexistence among groups of secondary networks: the opportunistic use of spectrum does not necessarily require prior agreement among the secondary networks intending to use the same TV frequency channel.

Although the IEEE 802.19.1 standard [22] defines methods and algorithms for co-existence between secondary cognitive networks, these have been deemed impractical due to the complexity of the architecture and the nature of fixed network management, for which adjustment of interference-reducing operational parameters is not obligatory. The IEEE 802.22 standard [23] indicates only that when no vacated TVWS channels are available for allocation, or when different groups share the TVWS channel for spectrum efficiency purposes, then "the A-BS may use a mechanism for proper resource sharing between the overlapping groups or the A-BS and some groups after channel switching" [23]. However, such a resource sharing mechanism is not detailed or defined by the standard.

# 1.2 Problem definition

Even though there are different methods and algorithms for coexistence between secondary cognitive networks, the complexity of the architecture and the non-obligatory adjustment of the operating parameters to achieve coexistence avoiding mutual interference have made it a challenge to enable the TVWS channel sharing. The lack of a mechanism for sharing the TVWS spectrum can lead to denial of service to networks due to no vacant TVWS channels available, especially considering the expected growth in the deployment of fixed and mobile secondary networks. Besides that, not enabling the TVWS channel sharing also causes inefficient spectrum use, a limited resource that provides high-impact services. Failure to promote the efficient use of secondary spectrum could lead to the same spectrum scarcity drawbacks as the exclusive-usage spectrum.

In the preliminary analysis to model the coexistence scenario without the use of a TVWS channel sharing mechanism, the results showed that the normalized channel capacity offered to a vehicular network decreases as the presence of fixed secondary networks on the TVWS channels increases. The normalized channel capacity offered to the vehicular network decreases from 0.6 to zero when the presence of the fixed secondary networks varies from 25% to 100%.

Besides that, not enabling TVWS sharing would force the vehicles to execute spectrum detection techniques until they find an available TVWS channel. When the vehicles find the available TVWS channel, the vehicles move their data transmissions to the new TVWS channel. If the selected TVWS channel becomes busy by a secondary network farther along the route, the vehicles shall rerun the process to find a new TVWS channel. Considering the increment in the deployment of secondary networks, the process of finding a TVWS channel for exclusive vehicular access may take around 0.2 s when the presence of fixed secondary networks is approximately 70%, and the TVWS channel sharing is not enabled.

## 1.3 Hypotheses

- 1. There are opportunities for vehicular access even when a fixed secondary network occupies the TVWS channel. The number of the channel access opportunities for vehicular communications could be estimated using parameters such as the probability of channel occupation made by the fixed secondary network, and the speed of the vehicles.
- 2. Vehicles may use these opportunities for transmission, instead of spending time looking for a new available TVWS channel. Sharing the TVWS channel with a fixed secondary network reduces the average channel access delay for vehicular communications in at least 10%.
- 3. Enabling the TVWS sharing among fixed secondary and vehicular networks increases the offered channel capacity at least 10%, in scenarios where active fixed secondary networks are along the route. The increment in the channel capacity is compared with the offered by the system when sharing the TVWS channel is not allowed, and the vehicular network must stop transmitting.

# 1.4 Objectives

# 1.4.1 General Objective

To design and model a resource sharing mechanism to enable the coexistence among fixed secondary networks and vehicular networks when no vacant TVWS channels are available for allocation, allowing resource sharing between the overlapping groups according to the IEEE standard 802.22. The effectiveness and level of service provided by the proposed mechanism will be evaluated using standard metrics and comparing with the IEEE 802.22 standard scenarios with no coexistence enabled among secondary networks.

# 1.4.2 Specific Objectives

- 1. To design a system model and a quantitative metric for establishing the feasibility of opportunistic channel access for vehicular communications over TVWS considering the primary TV users as well as other opportunistic secondary networks.
- 2. To design and model a resource allocation mechanism for vehicular communications on TVWS spectrum in scenarios considering the presence of fixed secondary networks and no vacated TVWS channels available for allocation.
- 3. To evaluate the performance of the resource sharing mechanism via numerical evaluations and simulations, considering metrics such as channel capacity, Jain's Index, spectral efficiency, and comparing with the IEEE standard 802.22 mechanism.

# 1.5 Contributions

The contributions of this thesis are:

- A feasibility study of the opportunistic access for vehicular networking considering the TV primary users and secondary users on the TVWS spectrum. This study presents a system model that integrates the technical requirements of primary users, and the conditions that must be met to allow secondary users on TV bands. The study analyzes the impact of vehicle speed and channel verification distance on the perception of channel occupancy and the identification of transmission opportunities over the TV band.
- The proposal of a new metric, the Channel Availability for Opportunistic Vehicular Access (CAFOVA) that relates the channel occupation policy over the White-Fi network, the vehicle's speed, and the channel verification distance.
- The proposal of a White Space Resource Sharing (WSRS) mechanism for vehicular networks based on power control and channel allocation. This mechanism is used when the fixed and vehicular networks are overlapped, no vacated TVWS channels are available for allocation, and both networks have to share the same TVWS channel.
- A thorough analysis of the impact of TVWS channel sharing between a vehicular network and secondary fixed networks in terms of the channel capacity offered to the vehicular network and attention to fairness and spectral efficiency of resource allocation.

# 1.6 Publications

## Journals

J1. A. Arteaga, S. Céspedes, and C. Azurdia-Meza, "Vehicular Communications over TV White Spaces in the Presence of Secondary Users," IEEE Access, vol. 7, pp. 53496–53508, 2019.

J2. A. Arteaga, S. Céspedes, and C. Azurdia-Meza, "Towards the Coexistence of Cognitive Networks for Vehicular Communications on TVWS for IEEE Std. 802.22", submitted to IEEE Transactions on Cognitive Communications and Networking, Sep 2021.

## International and national conferences

C1. A. Arteaga, P. Palacios, S. Céspedes and C. Azurdia-Meza, "Evaluation of Opportunistic Access Strategies for Vehicular Networking Over TVWS," 2020 IEEE 92nd Vehicular Technology Conference (VTC2020-Fall), 2020, pp. 1-5.

C2. A. Arteaga, S. Céspedes, and C. Azurdia-Meza, "Next Generation Vehicular Communications via Interworking of DSRC and TV White Space," in *Proc. 2018 18th International Symposium on Antenna Technology and Applied Electromagnetics*, Waterloo, Canada, August 2018.

C3. T. Lara, A. Arteaga, and S. Céspedes, "Comparison of Path Loss Models for Vehicular Communications in TV White Spaces," in *Proc. of the IV Spring School on Networks*, Valdivia, Chile, October 2018, 2178: 6-8.

C4. A. Arteaga, S. Céspedes, and C. Azurdia-Meza, "Next Generation Vehicular Communications via Interworking of DSRC and TV White Space," Poster in *VII Encuentro de Mujeres en Ingeniería - Chile WiC*, Valparaíso, Chile, October 2018, 2178: 6-8.

C5. A. Arteaga, S. Céspedes, and C. Azurdia-Meza, "Spectrum Sensing for Vehicular Dynamic Spectrum Access," in *Proc. of the II Spring School on Networks*, Santiago, Chile, November 2016, vol. 1727, pp: 8-13.

# 1.7 Outline of the thesis

The rest of the thesis is structured as follows:

- Chapter 2 presents the state of the art of the main topics related to cognitive vehicular networking, dynamic spectrum access for vehicular networking on the TVWS spectrum, the IEEE standard 802.22, and secondary opportunistic networks and coexistence on TVWS spectrum.
- Chapter 3 presents the general coexistence scenario for vehicular communications on the TVWS Spectrum, which includes the TV bands primary users, secondary networks, and vehicular networks. Then, the coexistence problem is focused on the resource sharing between IEEE 802.22 secondary networks. Finally, the White Space Resource Sharing mechanism is detailed.

- Chapter 4 presents the evaluation of the scenarios to study the coexistence for vehicular communications in the TVWS spectrum. The evaluation includes a feasibility study of TVWS spectrum sharing in the general scenario; a comparative evaluation of two strategies for opportunistic vehicular access over the TVWS spectrum in the presence of other opportunistic networks; and the performance evaluation of the White Space Resource Sharing mechanism for the coexistence according to the IEEE standard 802.22.
- Chapter 5 describes the main conclusions of this thesis and presents an outlook on future research regarding these topics.

# Chapter 2

# Literature review

This section presents the main topics and concepts of this thesis related to:

- Vehicular networking, applications, and communications technologies
- Cognitive vehicular networks
- Dynamic spectrum access for vehicular networking over TVWS
- IEEE standard 802.22
- Secondary opportunistic networks over TVWS and coexistence

## 2.1 Vehicular Networks

Traffic congestion, environmental pollution, and traffic accidents have driven research and development in the automotive industry, not only in the manufacture of cars but also in the use of technologies to improve safety on the road. According to the 2015 global status report on road safety provided by the World Health Organization (WHO) [24], the total number of victims by road traffic accidents has plateaued at 1.25 million per year; almost half of all road traffic deaths are among pedestrians, cyclists, and motorcyclists. Traffic congestion decreases the transport systems efficiency especially in urban areas, which harms the quality of life of the people. Exchanging information between the transport systems has been explored as an alternative to improve traffic efficiency [25] and road safety [26], which is known as Intelligent Transport Systems (ITS).

Vehicular networking provides the information exchange for vehicular ad hoc networks (VANETs) via vehicle-to-vehicle (V2V), vehicle-to-infrastructure (V2I), and vehicle-to-everything (V2X) communications which are fundamental features for the ITS deployment [27]. These communication modes allow offering different types of vehicular applications using different technologies. Some technologies are of general purpose such as the cellular network and WiFi, while others are designed specifically for vehicular communications. The following presents a more detailed description of the vehicular applications and the principal technologies used in vehicular networking.

### 2.1.1 Applications and services

Vehicular applications and services are into several categories depending on the purpose of the applications which are active road safety, efficient traffic coordination, and information and entertainment (infotainment) [5], or on a broader categorization of safety and non-safety applications [28]. The applications and services have minimum quality of service (QoS) requirements depending on the maximum latency, the minimum transmission frequency of the messages, the communication mode (broadcasting, cooperative, V2X, Internet access, among others.), and the required reliability. Table 2.1 shows the main vehicular applications and services for ITS, and the QoS requirements.

Safety applications: These applications seek to reduce the risk of traffic accidents through sharing safety-related information. Shared information could be related to vehicle's traveling state information (e.g., current position, real-time speed, and direction), or event-driven safety information (e. g., emergency vehicle warning, traffic condition warning, emergency electronic brake lights, among others.). Safety applications are used to avoid collisions between vehicles, to assist the driver in lane changes detection dangerous situations, and to maintain a safe distance and a constant speed for platoon-based driving. In general, these applications require a maximum delay of 100 ms and a message transmission frequency of 10 Hz.

**Non-safety applications**: These applications are focused on providing traffic management and infotainment support to enhance the comfort of the drivers and passengers. Traffic management applications seek to improve traffic flow, helping drivers reroute in traffic jam situations or improving the efficiency of the traffic lights schedule. Infotainment applications support location information services, entertainment, Internet access, among others. These applications can be considered delay-tolerant applications up to 500 ms, and they could demand a high data rate to download the information.

Use case	Category	Safety/Non-safety	Security/	Communication	Maximum	Minimum
		application	Reliability	mode	frequency	delay
Emergency	Active	Safety	High/High	Time limited periodic	10 Hz	100 ms
electronic break	road safety			broadcast on event		
lights						
Slow vehicle	Active	Safety	High/High	Periodic triggered	2 Hz	100 ms
warning	road safety			by the vehicle		
Vulnerable road	Active	Safety	High/High	V2X cooperative	1 Hz	100 ms
user warning	road safety			awareness		
Stationary vehicle	Active	Safety	High/High	Time limited periodic	10 Hz	100 ms
warning	road safety			broadcasting on event		
Traffic condition	Active	Safety	High/High	Time limited periodic	1 MHz	100 ms
warning	road safety			safety broadcasting		
				or authoritative		
				message triggered		
Lane change	Active	Safety	High/High	V2X cooperative	10 Hz	100 ms
assistance	road safety			awareness		
Overtaking vehicle	Active	Safety	High/High	V2X cooperative	10 Hz	100 ms
warning	road safety			awareness		
Pre-crash sensing	Active	Safety	High/High	Broadcast of pre-crash	2 Hz	100 ms
warning	road safety			state		
Regulatory/	Traffic	Non-safety	High/High	Authoritative message	1 Hz	N/A
contextual	management			triggered by traffic		
speed limits				management entity		
Intersection	Traffic	Non-safety	High/High	Periodic, permanent	2 Hz	100 ms
management	management			message broadcasting		
Traffic light optimal	Traffic	Non-safety	High/High	Periodic, permanent	2 Hz	100 ms
speed advisory	management			message broadcasting		
Cooperative flexible	Traffic	Non-safety	High/High	Periodic, permanent	1 Hz	500  ms
lane change	management			message broadcasting		
Electronic troll	Traffic	Non-safety	High/High	I2V broadcasting and	1 Hz	500  ms
collect	management			unicast full duplex		
				session		
Point of interest	Infotainment	Non-safety	Medium/	Periodic, permanent	1 Hz	500  ms
notification			Medium	message broadcasting		
Local electronic	Infotainment	Non-safety	High/High	Duplex communication	1 Hz	500  ms
commerce				between the RSU and		
				vehicles		
Media download	Infotainment	Non-safety	Medium/	User access to	1 Hz	500  ms
			Medium	Internet for download		
Map download and	Infotainment	Non-safety	Medium/	Access to Internet	1 Hz	500  ms
update			Medium	for download and update		

Table 2.1: Quality of service requirements of vehicular applications and services. Adapted from [26]

## 2.1.2 Communication technologies

V2V, V2I, V2X communications may be supported by several wireless technologies including general purpose technologies such as the cellular network and WiFi, through technologies designed specifically for vehicular communications such as the dedicated short-range communications (DSRC), or using novel technologies such as millimeter waves. Each technology has different characteristics regarding the coverage range, the data rate, the channel access mechanism, among others; these characteristics may benefit or harm the applications' performance considering the minimum QoS requirements compliance. The following presents a more detailed description of the traditional and novel communication technologies used in vehicular networking.

#### DSRC

DSRC is a technology designed to support a variety of applications in V2V and V2I scenarios in the 5.9 GHz band, mainly to enable road safety applications [6]. Each DSRC-equipped vehicle has an on-board unit (OBU) to broadcast its safety messages with information such as speed, location, and acceleration, and to receive the safety messages from the neighbors. In safety applications, the vehicle uses the received safety messages to calculate the trajectory of other vehicles and to avoid possible collisions with other vehicles. Vehicles may also communicate with DSRC roadside units (RSUs) which are infrastructure equipment installed on the route. The RSU may communicate higher level information obtained by the processing the messages sent by several vehicles. The coverage of the OBU is up to 300 m, and the RSU is up to 1 km.

Due to the requirements of road safety applications, several countries have assigned exclusive spectrum to DSRC. The DSRC spectrum in Europe is 70 MHz (5.855-5.925 GHz), and 80 MHz (5.770-5.850 GHz) in Japan. There is a 5 MHz band guard at the beginning of the DSRC band, and each channel is 10 MHz but pairs of channels can be combined into a 20 MHz channel; the CCH is in the middle of the six SCH as shown in Figure 2.1.



Figure 2.1: DSRC frequency band allocation. Source: Own elaboration based on [6].

The IEEE 802.11-2012 standard defines the DSRC physical layer [29]. The layer is divided into two sub-layers: the physical medium dependent (PMD) sub-layer and the physical layer convergence procedure (PLCP) sub-layer. The PMD is connected to the wireless medium using orthogonal frequency division multiplexing (OFDM), while the PLCP defines the mapping between the MAC frame and the OFDM symbol. Because the DSRC channels are 10 MHz, the PHY layer uses the 10 MHz OFDM channel parameters shown in Table 2.2. Table 2.3 shows the data rate options in DSRC 10 MHz OFDM channel depending on the modulation technique, the code bit rate, and the coding rate. Selecting a specific modulation scheme and the coding rate affects the achieved data rate, and the minimum sensitivity required to decode a received frame [29].

Parameter	Value
Number of data subcarriers	48
Number of pilot subcarriers	4
Subcarrier frequency spacing	$156.25 \mathrm{~kHz}$
Guard interval	$1.6 \ \mu sec$
Symbol interval (including guard interval)	$8 \ \mu sec$

Table 2.2: 10 MHz OFDM channel parameters for IEEE 802.11p [6]

Table 2.3: Data rate options in a DSRC 10 MHz OFDM channel [6]

Modulation	Coded bit rate	Coding rate	Data rate	Minimum sensitivity
technique	$({ m Mb/s})$		$(\mathrm{Mb/s})$	(dBm)
BPSK	6	1/2	3	-85
BPSK	6	3/4	4.5	-84
QPSK	12	1/2	6	-82
QPSK	12	3/4	9	-80
16–QAM	24	1/2	12	-77
16–QAM	24	3/4	18	-73
64–QAM	36	2/3	24	-69
64–QAM	36	3/4	27	-68

The DSRC data link layer is also divided into two sub-layers: the medium access control (MAC) sub-layer responsible for sharing the wireless medium, and the logical link control (LLC) that defines the data transfer over the channel. Medium access in 802.11p is carrier sense multiple access/collision avoidance (CSMA/CA) where the sender can send its message if the channel is idle. If the medium is busy, the sender must wait a number of idle time slots according to a random backoff process. Using CSMA/CA for access channel increases the number of collisions in high vehicular density scenarios, especially for vehicular applications based on broadcasting.

One aspect to consider when using DSRC for vehicular networking is the unbalance between the links due to the difference in the OBU and the RSU coverage range. The reliable communication range from the RSU to the OBU is up to 1,000 m, while the communication range from the OBU to the RSU is up to 300 m. In this case, the OBU is capable of receiving the RSU message when the vehicle is into the RSU coverage, but the RSU only receives the OBU messages when the OBU is close enough to cover the RSU. This situation should be considered in the application design and implementation because the QoS requirements compliance could be affected, especially the maximum delay. When the application requires a shorter response time or a lower delay, technologies such as LTE with a higher coverage range can be used.

#### LTE-V

LTE-based V2X services have been specified by the Third Generation Partnership Project (3GPP) to take advantages of widely deployed LTE networks and the low cost of the user equipment (UE) [30]; the standard is commonly referred to as LTE-V, LTE-V2X, or cellular V2X. Release 14 LTE supports one-to-many communications that are useful for broadcasting, and the device-to-device communications through the PC5 interface to enable V2V links without routing via the base station (the evolved NodeB or eNodeB) [31]. For V2I communications, LTE Release 15 enables the access network (Uu) interface to establish communication between the vehicle and the eNodeB. Release 15 that will support fifth-generation (5G) V2X communications in under discussion. New use cases will be considered in Release 15, including autonomous driving, platooning, sensor and map sharing, information sharing for partial/conditional and high/full automated driving, and remote driving, among others [32]. A comprehensive study of LTE-V V2V communications that shows its potential as an alternative to DSRC is shown in [33].

#### WiFi

This technology is a popular option to provide wireless connectivity because there are WiFi hotspots deployed in urban areas, the WiFi access is low-cost, there is a very high number of devices with WiFi interfaces, and the WiFi technology can provide data rates up to 54 Mb/s. However, the limited coverage range of the WiFi access point and the high mobility of vehicles causes unstable connections. For this reason, WiFi has been considered as an offloading technology more than the main technology for vehicular networking [34]. Each vehicle can get a short duration to connect to a WiFi hotspot depending on the vehicle mobility to send short messages, or to download a relatively small data volume of the Internet [35].

#### Novel technologies

Considering that the amount of spectrum for vehicular communications is limited and that it is mainly intended for security applications, other emerging technologies have been explored to extend the available spectrum for vehicular networking. Among the technologies explored are the millimeter waves, light fidelity (LiFi), and TV White Spaces (TVWS).

Millimeter waves (mm-wave): this spectrum corresponds to the 60 GHz frequency band, the operation of anti-collision radars and other short-range ITS applications in V2V scenarios is allowed [36, 37, 38], following the IEEE 802.11ad standard [39]. This technology offers a bandwidth channel up to 2.5 GHz in the 59-66 GHz band and a coverage range of a few meters, increasing the available bandwidth for short-range vehicular applications.

Light-Fidelity (LiFi): It is a high speed bi-directional wireless technology that uses the visible light to transmit data, exploiting the visible light unlicensed spectrum [40]. This technology could be used in short-range applications by sending messages through light-emitting diode (LED) bulbs and a photo-diode. Prototypes have been presented to send a warning message from the front vehicle to the rear vehicle when the front vehicle slows down the speed [41]. Enhanced prototypes with LED transmitter and a camera receiver for experiments under real driving conditions is presented in [42], and visible light communications (VLC) systems using LED-based traffic lights has been explored for ITS deployment [43].

**TV** white spaces (**TVWS**): These are the available channels in TV frequency bands (470 MHz-806 MHz). Among the reasons why TVWS may exist in a specific area are: a) there are no TV transmitters using the channel in the area of interest; or b) there is a TV transmitter using the channel in the area of interest, but the TV emission is OFF during specific hours (e.g., at night). Spectrum regulators in several countries allow fixed and portable/mobile secondary users in the TV bands with the condition of protecting the TV primary user against harmful interference. There are several studies about using TVWS in vehicular networking because the TV band propagation characteristics allow a greater transmission range than that achieved with DSRC [44, 45, 46, 47, 48, 20, 49, 50]. The TVWS network has been proposed as the offloading technology when there is degradation in the DSRC channels, as the backup network for sending control messages in V2V and V2I scenarios, and as the principal network for Internet access for infotainment services. The availability estimation and the use of the TVWS for vehicular networking will be addressed in detail in Section 2.3 related to dynamic spectrum access for vehicular networking.

## 2.2 Cognitive Vehicular Networks

Several technologies were explored to allow VANETs communication through cellular or WiFi networks, while the automotive industry developed the DSRC technology for V2V and V2I communication [6]. As previously mentioned, the Federal Communications Commission (FCC) in the U.S. allocated 75 MHz of licensed spectrum in the 5.9 GHz band for DSRC communication [51] divided in 7 channels of 10 MHz each, and 5 MHz guard band at the beginning of the DSRC band. However, DSRC channels could become congested in scenarios of high traffic density, which impairs the performance of the applications in situations of high vulnerability for drivers and passengers. Considering there are ITS applications with different bandwidth and maximum tolerated delay requirements, resource allocation within DSRC channels and the radio spectrum scarcity must be considered in the VANETs approach. Taking into account the different technologies that can be used for the deployment of vehicular communications (see Section 2.1.2), cognitive radio in vehicular networking has been explored to extend the available spectrum for vehicular applications combining the access to several technologies such as DSRC, WiFi, LTE, and TVWS [52, 53]. Software-defined radio allows to implement OBUs and RSUs based on Cognitive Radio [54], which is a paradigm to use software routines to adjust the operational parameters of the devices (e.g., the active wireless technology, the modulation scheme, the transmission power, etc.) according to the current network environment.

Vehicular networks based on cognitive radio allow integrating multiple technologies to implement a heterogeneous network with extended capabilities, and to provide a flexible network to react to different communication requirements according to changes in the vehicular traffic. Network features such as beaconing, routing, and message dissemination may be adapted according to the channel congestion, the vehicular density, or the application requirements. This paradigm is used to design the architecture for the interworking between technologies because it facilitates the data collection, the decision making, and the configuration of the network elements. Flexibility could increase implementing the network through software-defined networking (SDN), a paradigm based on the separation of the control plane and the data plane [55]. SDN has been explored to improve the vehicular network capabilities through scalability, flexible resource allocation, and advanced network management using SDN controllers that sent configuration settings to SDN switches in the OBU and the RSU [56, 57, 58, 59]. This approach allows study alternatives to solve some challenges related to messages dissemination [60], service QoS guaranty in mobile environment [61], the integration with cloud and fog computing [62] and the optimization of resource allocation in heterogeneous VANETs (HetVNETs) [63, 64].

# 2.3 Dynamic Spectrum Access for Vehicular Networking

Channel availability is a fundamental requirement for the successful operation of vehicular applications, but the lack of available spectrum for wireless technologies is a threat for current and future applications designed for ITS. Theoretical analysis and simulations have shown that the DSRC spectrum is insufficient for the reliable transmission of the messages in safety applications, especially in high vehicular density scenarios when the packet delivery ratio decreases [65, 66, 67]. Since the DSRC spectrum is used mainly by the safety applications, it is necessary to define what spectrum will be available for the applications such as video streaming or file downloads, which demand a high data rate and bandwidth. There is a spectrum for the vehicular application and the increment in the number of vehicles on the roads.

As an alternative to face the radio spectrum scarcity problem, not only for vehicular networking but also in other networks, dynamic spectrum access (DSA) has been explored. This approach allows unlicensed users to use available licensed frequencies with the condition of avoiding harmful interference to licensed (primary) users [13]. Several primary services have been reviewed by spectrum regulators to determine which services better support the presence of secondary users, according to the spectrum occupancy rate, the tolerance to interference, the band regulation, and the international coordination agreements for the service operation. Reports on the use of radio spectrum show there is low occupancy in the frequencies allocated to the television service (470 MHz-806 MHz) [68]. Moreover, some frequencies will become vacant due to the transition from analog to digital TV [69]; the available channels in TV frequency bands are known as TV White Spaces (TVWS) (see Section 2.1.2). Spectrum regulators such as the FCC in the U.S[70], and the CEPT [71] in Europe defined the technical requirements to allow secondary devices over the TV bands, in order to guarantee the protection of the primary user against harmful interference.

The FCC defines three types of secondary devices over TVWS [70]:

- Fixed: The power transmission is up to 4 W. It is used to bring wireless internet access in rural areas through TV bands. A fixed device must consult the geolocation database to determine the list of available channels and the position of the device. Also, a spectrum sensing technique can be used to detect wireless microphones or other allowed signals in TV bands. Fixed devices must communicate between them using the following frequencies: 54-60 MHz, 76-88 MHz, 174-216 MHz, and 470-512 MHz.
- Portable: The power transmission is up to 100 mW, and it is implemented as a wireless interface or an access point. There are two types of portable devices:

- Mode-I: a portable device that gets the spectrum occupancy information via a fixed device or a portable mode-II device.
- Mode-II: a portable device that can connect directly with the geolocation database to get the spectrum occupancy information, or using a spectrum detection technique. It must check the spectrum occupation every time it moves 100 meters.

The CEPT defines three types of secondary devices which are [71]:

- Personal or portable: a small device that can be transported by people as a mobile device, or it can be embedded in computers. It can be used inside the house, on the street or in transportation systems. It must have an Internet connection, allowing communication with other secondary devices.
- Home and office device: Devices such as television, video console, among others. It could be used in homes and offices but cannot be transported by people. It is useful to offer services that require a greater bandwidth or greater processing capacity than personal devices.
- Private and public access points: A device that provides WiFi service but with greater coverage, taking advantage of the propagation characteristics of the television band.



Figure 2.2: General architecture for vehicular dynamic spectrum access. Source: Own elaboration.

According to the classification of the secondary devices defined by the FCC and the CEPT, the cognitive RSU must follow the specifications of the fixed device (private/public access point according to the CEPT), while the cognitive OBU must be a portable mode-II device (a portable device according to CEPT). In the literature, dynamic spectrum access for vehicular networking is also known as vehicular dynamic spectrum access (VDSA) [72]. Figure 2.2 shows the general architecture for VDSA. The information about TV channel allocation is stored on a geolocation database accessible by the Internet. Cognitive OBUs can check the spectrum occupation locally using spectrum detection technique, or via the cognitive RSU.

### 2.3.1 Available spectrum estimation mechanisms

Secondary devices interested in the use of TVWS must ensure the channel is available before to transmit and must be ready to release the channel when a primary user requires it. There are two methods to determine the channel availability: a using a geolocation database with information of the channel allocation or b using a spectrum detection technique to recognize the presence of an active signal on the channel. The following presents the details of both methods.

#### Geolocation database

This database contains information of the incumbent services on TV bands, which are TV broadcasting, private land mobile service (PLMS), commercial mobile radio service (CMRS) on channels 14-20, offshore radiotelephone service, and low power auxiliary service (wireless microphones) [73]; the database also has information about the protected TV receive sites that are outside the protected contour of the TV stations. The FCC in the U.S. and Ofcom in U.K. offer access to the white space database through database administrators such as Google, LStelcom, and Spectrum Bridge, to obtain the lists of available channels. A fixed or portable mode-II device must consult the database periodically using its geolocation capacities and an Internet connection, while a portable mode-I device must use a fixed or a portable mode-II device. The secondary device must use a non-TV channel technology to connect to the geolocation database (e.g., a cellular connection, a WiFi connection or a fixed connection), because this connection must always be available to consult the database.

The use of geolocation databases has been broadly proposed for VDSA [44, 45, 46, 74] because it is easily implementable and because the allocation of the TV channels remains stable. If a channel is assigned to a TV transmitter, the channel will be busy while the TV transmitter is operating. However, the database dependency for enabling V2V communications over TWVS could be unsuitable due to the additional latency required for accessing the database, not to mention that it is not clear where the geolocation database should be stored to guarantee that vehicles always have access to it, especially if there is no connection to the infrastructure.

#### Spectrum detection techniques

The spectrum detection techniques allow obtaining the real-time status of the channel occupation because they detect if the channel is idle, or if there are incumbent or a secondary users on the channel. There are several spectrum sensing techniques for the available spectrum estimation, but the complexity of a detection technique depends on a) the knowledge of the transmitted signals according to features such as the modulation scheme, the packet format, the pulse shape, among others, b) the signal processing required over the reception, and c) the number of receivers required to process the potential primary signal. The more known information about the potential primary signals, the more reliable is the spectrum detection process.

In DSA for vehicular networking, spectrum detection techniques have been proposed as a way to check channel status locally, in particular when the access to a geolocation database is restricted by the nature of the ad hoc network, especially in V2V scenarios. In the following,

we revisit three of the most frequently used spectrum detection techniques employed in opportunistic spectrum detection [75], and discuss how to apply the techniques to VDSA.

**Energy detection**: This is a simple spectrum detection technique because of its low computational cost and low implementation complexity [76]. A signal is detected by comparing the energy in the channel with a threshold  $\lambda_E$  [dB], which in its simplest definition depends on the noise floor. If the energy in the channel exceeds  $\lambda_E$ , it means that a user is occupying the channel; otherwise, the channel is available to a secondary user. If y(n) is the received signal, its energy is defined as:

$$E = \frac{1}{N} \sum_{n=0}^{N} |y(n)|^2, \qquad (2.1)$$

where N is the number of samples of the signal. The threshold definition is an essential aspect for achieving an effective detection if there are low Signal-to-Noise ratio (SNR) conditions in the channel. The receiver does not need a priori knowledge of the primary user signal, but the accuracy of detection is affected if SNR is low. Improved versions of this technique are proposed for VDSA, where cooperative sensing is one alternative to increase the detection reliability in highly mobile scenarios, where several detection measures are combined to avoid missing a detection or producing a false alarm that prevents the use of available channels [77, 78, 79].

In [80], the authors present a TV spectrum measurement campaign to analyze the availability of vacant channels from a vehicular perspective. Also, several experimental V2V studies have been conducted to analyze the feasibility of the use of TVWS for vehicular communications [81, 82, 45].

**Cyclostationary analysis:** This technique is based on the cyclostationary characteristics of the received signal, which are modelled by its periodic behavior with period  $T_0$ , the mean value  $M_x(t)$ , and the auto correlation function  $R_x(t,\tau)$ , calculated in (2.2) and (2.3), respectively, as follows [83]:

$$M_x(t) = M_x(t + T_0), (2.2)$$

$$R_x(t,\tau) = R_x(t+T_0,\tau+T_0).$$
(2.3)

With this technique, it is possible to differentiate a primary signal from noise, since the noise is a random process and its autocorrelation function is not periodic. The detector calculates the autocorrelation function of the received signal and compares it with a known value  $\lambda_C$ , which can be the cyclic frequency of the signal [84]. The technique is considered of medium complexity because is robust to noise uncertainty, but a priori knowledge of the primary signals is required. Performance of this technique does vary depending on channel noise. If the channel noise is stationary, the performance of this technique is low; nevertheless, this technique performs well when the SNR is low because  $\lambda_C$  does not depend on the signal level, but on the periodic behavior of the signal. However, a long sensing time is required for accurate detection results, which is inconvenient when vehicles move at high speeds since

sensing data may become outdated very fast. Studies in [85, 86] show the performance of the cyclostationary analysis in fading channels, with similar conditions to the ones expected for vehicular communications.

Waveform Based Detection: This technique is applicable when some of the signal patterns are known (e.g., preamble, spread spectrum sequence, pilot signal, etc.). Identifying a primary user requires that the receiver calculates the correlation between one of the known patterns of the received signal and a bank of known patterns of the possible primary signals. If y(t) is the received signal, s(t) is a known signal, and w(t) is the white Gaussian noise, the correlation between both signals is calculated as follows [75]:

$$M = R_{\rm e} \left[ \sum_{n=0}^{N} y(n) s^{*}(n) \right], \qquad (2.4)$$

where \* is the conjugate. Correlation when the primary user is present or absent can be expressed according to (2.5) and (2.6), respectively:

$$M = \sum_{n=0}^{N} |s(n)|^2 + R_{\rm e} \left[ \sum_{n=0}^{N} w(n) s^*(n) \right], \qquad (2.5)$$

$$M = R_{\rm e} \left[ \sum_{n=0}^{N} w(n) s^{*}(n) \right].$$
 (2.6)

M is compared with a  $\lambda_W$  value to determine if the correlation is high, indicating there is a primary user in the channel; if the correlation is low, it means there is only noise in the channel. The complexity of waveform-based detection is high because in some cases it needs to demodulate the signal to obtain the preamble, so the cognitive device must support the processing of all possible primary signals.

Waveform-based detection can rely on the mechanisms used by the primary communications systems to correct possible errors and use corrected information for detection when the SNR is low. This technique is similar to a matched filter with a perfect knowledge of the primary user; before sensing, features such as channel bandwidth, modulation scheme, or packet format are required. The sensing time is low because the features are known in advance, but the reliability of the technique depends on the level of knowledge of both users and signal patterns. The technique has been evaluated for fading channels, showing that time variations introduce degradation in the detection performance [87].

Due to the low latency in the execution of the energy detection technique, this could be useful in vehicular scenarios to allow each vehicle to check channel status across the route, reducing the latency of continuous access to the geolocation database due to the rapid changes of location of the moving vehicles. Cooperative sensing is one alternative to increase the detection reliability in highly mobile scenarios, where several detection techniques are combined to avoid missing a detection or producing a false alarm that prevents the use of available channels [77, 78, 79].

#### 2.3.2 Spectrum estimation based on the infrastructure

One approach for the available spectrum estimation is using an infrastructure-based architecture, where the spectrum occupation information is stored in the elements of the network as proprietary servers, or via Internet. The proposed architecture presented by Matinmikko *et al.* [88] will be used as an example to define the elements that can be used for an estimation of the available spectrum based on infrastructure.

The proposed architecture to manage the spectrum occupancy information in [88] uses four components: the database, the cooperative spectrum sensing system, the pilot channel, and the decision component. The database can be like the geolocation database presented in Section 2.3.1, which has information about the primary services on the TV band. The cooperative spectrum sensing system is a set of cognitive radio nodes executing a spectrum detection technique such as energy detection, correlation-based detection, or waveform based detection (see Section 2.3.1), collecting the information about the real-time channel status. The results of all the nodes are combined and stored on a database as sensing results. The pilot channel is a dedicated TV channel to announce the status of each channel on the band.

The decision component has three stages: the first stage is selecting the spectrum estimation tool according to the scenario: if the channel occupation made by the primary users change frequently, a spectrum detection technique such as energy detection is the best option; using the database could cause to make the decision using outdated occupancy information. The second and third stages are applied if a spectrum detection technique is selected. In the second stage, the measurement procedure and the parameters of the technique are defined such as the sensing duration, the channels to be sensed, the thresholds, among others. In the third stage, the cooperative sensing scheme combines the sensing results of the cognitive radio nodes to make the channel occupation decision. The spectrum availability information is announced to the user device via the pilot channel.

The architecture for dynamic spectrum access based on infrastructure is commonly used for V2I scenarios, where the infrastructure is responsible for obtaining the spectrum occupancy information and to attend the spectrum request of the vehicles via the RSU. In [44] and [46], the geolocation database is used as the only spectrum availability estimation tool because the studied problem assumes that the information of the TV channels does not change frequently. In [44], the vehicle keeps communication with the RSU to obtain the list of the available channels in the vehicle location, and the control of the spectrum allocation is centralized to exploit the spectrum more efficiently. The proposed infrastructure has a geolocation database server, the RSUs installed on the road, and an application server for the infotainment content. In [46], the authors propose an Internet-based TVWS infrastructure for media streaming services, as known as White-Fi Infostation [74]. The infrastructure includes the Geolocation Database Server (GDBS), the Registered Location Secure Server (RLSS), and the fixed White-Fi Infostations. The GDBS make the query of the available TVWS to the geolocation database via Internet, and the RLSS coordinates the optimal spectrum utilization for the White-Fi Infostations. When a TVWS is allocated for a vehicle, the multimedia content is transmitted using the TVWS. Other studies that use the available spectrum estimation based on infrastructure are the database approach are [44, 45, 46, 74, 80, 89].

#### 2.3.3 Spectrum estimation based on the vehicles

The estimation of the available spectrum based on vehicles has been studied for V2V scenario where there is no intervention of infrastructure elements such as the RSU. In this approach, the OBU is a portable mode-II secondary device that implements a spectrum detection technique to check the channels occupation locally. Other approaches because it can access to geolocation database directly using its location capabilities, and it can implement a spectrum detection technique.

In [48], the performance of energy detection over a Rayleigh fading channel is analyzed, using the average probability of missing detection which is the probability that a primary user is not detected because of a high threshold in the detection technique. Energy detection on fading channels is also analyzed in [90, 91]. Experiments of energy detection for V2V communications are presented in [80, 92, 81]. In [80], spectrum measurements are collected using a conventional spectrum analyzer when a vehicle travels between Boston, MA and West Stockbridge, MA at an average velocity of 96 km/h. Results show that energy values increase near to a TV transmitter and the energy levels decrease in rural areas far away from an active TV transmitter. In [92], a prototype of a cognitive radio device for energy detection on board of a vehicle is presented. The prototype implements the control channel and the data channel on TV band using spectrum sensing and distributed coordination functions on an SDR device. In [81], another secondary device prototype is used to transmit a video from a vehicle to another using the data channel over a TVWS. When an emulated primary user is detected, the secondary device must change to a new data channel according to the coordination function decisions.

Spectrum detection made by an OBU is useful for V2V communications because a local spectrum estimation can be done. One advantage is that mobility effects can be considered when available channels are searched. For the geolocation database, the FCC recommends that a mobile device must consult the channel availability when the device moves 100 meters, (e.g. every 3 seconds if the vehicle travels up to limit speed of highway in the USA which is 120 km/h.) A drawback of spectrum detection based on vehicles is there is not possible to have centralized control of spectrum allocation, then there is no guaranty of efficient spectrum usage. Also, it is not possible to get priority treatment for the transmission when its connection is lost. Another aspect to consider is the device must stop its current transmission to make spectrum detection in case of having only one transmitter/receiver, which reduce the effective data throughput. Also, there must be a mechanism to coordinate the channel tuning between the vehicles to establish communication between them. This coordination has been explored through the rendezvous protocol that defines sequences of channel jumps to get a transmitter and receiver to tune the same channel as many times [93, 94]. This approach requires increasing the number of successful meetings and decreasing the percentage of failed meetings, ensuring the establishment of communication between vehicles.

#### 2.3.4 Resource allocation mechanisms

When the available opportunistic spectrum has been identified, the next step is to decide how to assign it when there are several interested users. In DSA, the channels that will be assigned correspond to those that are not being used by TV transmitters which varies from one area to another. It must be considered in VDSA because the vehicles will use an available TVWS at one location, but the same TVWS may be busy a few kilometers later along the route. The busy TVWS may be being used by an active TV transmitter, or by other secondary users such as IEEE 802.11af or IEEE 802.22 networks. Spectrum must be assigned according to some criteria defined by the network, considering aspects such as a the efficiency of spectrum utilization, b the interference reduction, c the traffic's QoS requirements, among others. The evaluation of the criteria and the resource allocation decision can be done using centralized, distributed, or cluster-based approaches:

**Centralized:** A central node collects the information about spectrum occupation and users requirements to make decisions about channel allocation. In VDSA, the information about spectrum occupation may come from the geolocation database, or the cognitive RSUs that detect the spectrum occupation using spectrum detection techniques, or both (see Section 2.3.2). Centralized resource allocation has several advantages including the efficient spectrum usage because there is a controlled allocation of the resource to avoid the interference, and to know in a broad way the network status and the users' requirements. However, the centralized approach requires to mitigate the possible failures in the central node, to avoid that the resource allocation process stops working if the central node fails.

**Distributed:** Each node makes decisions based on its information or by cooperation with neighboring users. In cooperative approaches for VDSA, each vehicle detects one or more TVWS and share the information with near vehicles to decide if the TVWS is busy or available. Each vehicle selects an available TVWS, or a cooperative mechanism could be used to assure that several vehicles choose the same TVWS (see Section 2.3.3). This approach has advantages such as lower signaling overhead, and faster decision time, among others. As to drawbacks, there is no efficient spectrum use because each vehicle chooses an available TVWS without considering the global spectrum occupation, and there is no guarantee that interference will not occur between users and other secondary users.

**Cluster-based:** This is a hybrid between centralized and distributed approaches in which the nearby nodes are grouped into clusters, and the cluster head is responsible for collecting and combining the sensing information from all the nodes in the cluster [95]. The cluster head provides the spectrum allocation vector for the cluster; in case the cluster head fails, another node must assume the cluster head role. The spectrum allocation scheme for each cluster may be shared with a centralized node to get the global spectrum occupation on the vehicular network.

For modeling the resource allocation problem for VDSA considering other secondary users, the author in [49] proposed a cognitive vehicular network where the road is divided into segments with an RSU in each segment that uses the geolocation database and spectrum sensing techniques to estimate the TVWS occupancy. The vehicles must register with the RSU of the current segment to send their channel allocation request and to obtain channel allocation results. This proposal divides the time into equals scheduling cycles, and RSU allocates channels to vehicles at the beginning of every scheduling cycle. This approach assumes that the primary user tolerates a maximum collision probability, and defines the transmission time of the vehicle on a TVWS depending on the data rate, the number of packets vehicle tends to transmit in the current schedule, and the maximum allowed scheduling time on the channel. To maximize the throughput for all the vehicles, these are classified in priority classes to schedule a channel for higher priority vehicles earlier than lower priority vehicles. This scheduling for vehicular communications make flexible three constraints related to meet the FCC's requirements on the total transmission power on a TVWS channel; other to provide fairness among the vehicles, and other to avoid mutual interference among the vehicles. This resource allocation mechanism is proposed for the coexistence between 802.22 networks and a cognitive vehicular network: the vehicular network knows the 802.22 upstream scheduling information and sends burst traffic during the idle slots on the 802.22 upstream.

Another approach to the resource allocation task in VDSA is to achieve optimal TVWS network capacity considering to reduce or avoid the co-channel and adjacent channel interference in a White-Fi network (i.e., IEEE 802.22/802.11af network) form by White-Fi access points and vehicles. Authors in [96] propose a define the configuration profiles for each TVWS and a set of transmission power configuration profiles. Obtaining the joint channel-power selection profile is possible to obtain the optimal channel/power selection profile for all the White-Fi access points in order to maximize the network throughput.

There are other approaches to resource allocation for general cognitive radio network based on the economic perspective of the secondary use of licensed spectrum [97], but this approach will not be considered in this thesis proposal.

## 2.4 IEEE standard 802.22

The IEEE standard 802.22 uses TV band frequencies (54-862 MHz) with 6, 7, or 8 MHz channel bandwidth. The standard defines two PHY operation modes (i.e., PHY-OM1 and PHY-OM2) to support the basic and advanced 802.22 systems. The PHY layer mechanism is Orthogonal Frequency Division Multiple Access (OFDMA) with up to 2,048 subcarriers divided into data sub-carriers, pilot sub-carriers, guard, and null sub-carriers. Binary phase-shift keying (BPSK) up to 64-bit Quadrature Amplitude Modulation (64-QAM) schemes are supported; Table 2.4 lists additional 802.22 PHY features.

The Medium Access Control (MAC) protocols provide functions for the protection of TV band incumbent services and self-coexistence between WRANs. The MAC supports unicast, multicast, and broadcast services, implementing a combination of access schemes that control contention between users, including polling, to simplify the access operation.

The A-WRANs provide all the functionalities of PHY, MAC, and cognitive radio technologies defined in the previous version of the standard, with the latest version including an additional PHY Operational mode PHY-OM2 to provide multi-hop relay operations using a Relay CPE (R-CPE), multiple channel operations, advanced security for regional monitoring applications, and enhanced broadband services. Multi-channel operation and multi-hop relay are optional features to increase network capacity. Therefore, although originally defined as a fixed communication system, in its most recent version, the IEEE 802.22 standard
Parameter	Specifications PHY-OM1	Specifications PHY-OM2
Frequency range	54-862 MHz	54-862 MHz
Channel bandwidth	6, 7, or 8 MHz	6, 7, or 8 MHz
Data rate	4.54 to 22.69 Mb/s	$3.61$ to $18.05$ up to $25.27~\mathrm{Mb/s}$ for
(6 MHz case)		SISO and single channel operation.
		57.77 to 288.85 up to 404.39 Mb/s for 4
		stream MIMO and 4 channel operation.
Payload Modulation	QPSK, 16-QAM, 64-QAM	QPSK, 16-QAM, 64-QAM, 256-QAM
Transmit EIRP	4 W maximum for fixed CPEs and BSs.	4 W maximum for fixed CPEs and BSs.
	100 mW maximum for mobile/portable	100 mW maximum for mobile/portable
	CPE (USA regulatory domain)	CPE (USA regulatory domain)
Multiple access	OFDMA	OFDMA
FFT Size	2048	1024
Cyclic prefix modes	1/4, 1/8, 1/16, 1/32	1/4, 1/8, 1/16, 1/32

Table 2.4: IEEE 802.22 PHY and MAC Parameters [23]

also has the potential to provide connectivity to Intelligent Transportation Systems with cost-effective, extended coverage [98].

Figure 2.3 illustrates an advanced base station (A-BS) providing broadband services for an advanced customer premise or portable equipment (A-CPE) through direct connectivity to the A-BS or through a relay CPE (R-CPE). A large number of A-CPEs may be connected to the A-BS, and several A-CPEs form a group managed by an R-CPE. An A-WRAN supports fixed and mobile (or portable) groups, depending on the mobility of the R-CPE. The R-CPE of a fixed group may be installed on top of a building, a house, or a tower, among others. The R-CPE of a mobile (or portable) group may operate in a vehicle, with other mobile users including cyclists and pedestrians, or even in animals for tracking purposes.

The cognitive radio capabilities in the 802.22 standard provide the mechanisms for protection of incumbents and the efficient operation of A-WRANs. The cognitive plane provides the spectrum sensing function (SSF), the geolocation (GL) function, the spectrum manager/spectrum sensing automaton (SM/SSA), and a dedicated security sub-layer 2. The SSF implements spectrum sensing algorithms, and the GL module provides the information to determine the location of the IEEE 802.22 device (A-BS or A-CPE).

#### Spectrum Manager (SM)

The SM plays a critical role in the network architecture. The SM maintains spectrum availability information, manages channel lists and quiet-period scheduling, and implements coexistence mechanisms. The SM also takes requests from the MAC/PHY and functions as the central point in the A-BS cognitive plane because it gathers all the spectrum availability information resulting from the spectrum sensing function and the database service. Using this information, the SM provides information to the MAC to remotely configure all the registered A-CPEs.



Figure 2.3: Advanced Wireless Rural Area Network (A-WRAN). Source: Own elaboration based on [23]

#### Spectrum Sensing Automation (SSA)

This entity is present in the A-BS and the A-CPEs. It independently implements specific procedures for sensing the RF environment at the initialization of the A-BS and before registering an A-CPE with the A-BS. The CPEs power up by first searching all of the channels in the area to see if an A-BS is present. Once an A-CPE locates an A-BS, authentication and connection setup are done. In addition, the standard requires incumbent-user sensing measurement and detection. The SSA at the A-CPE includes features to allow the A-CPE operation when it is not under the A-BS control. When the A-CPE is idle, the SSA conducts out-of-band sensing and reports to the BS to update the status of the channel list.

#### Geolocation Function and TV Bands Database

The A-BS SM communicates with the TV bands' database service to get a set of available channels in its location. The SM generates the available channel list using only those channels indicated for every device on the network. Each A-BS has a list of backup and candidate channels. A backup channel is one that has been verified as available; therefore, its use does not affect an incumbent service. A candidate channel is a channel that, once cleared from potentially harmful interference to incumbents, may become a backup channel.

# 2.5 Secondary opportunistic networks over TVWS and coexistence

Opportunistic usage of the available spectrum in TV bands have been explored as an option to deal with the spectrum scarcity caused by the growth in the devices connected to the Internet via wireless technologies, and by the high bandwidth requirement of the applications. IEEE has published two related standards for wireless communications via TVWS as a technology that extends the wireless coverage due to propagation characteristic in the VHF and UHF bands. The IEEE 802.11af standard is the specification of physical and medium access control layers for wireless local area networks employing TV bands [99], whereas the IEEE 802.22 standard specifies physical and medium access control layers for cognitive radio networks over TV bands [100]. IEEE 802.11af and IEEE 802.22 networks have been proposed to improve broadband access in rural areas [15, 16], to provide connectivity in education initiatives [17, 18, 19], to support agricultural projects [21], and to provide connectivity in emergency response plans [20]. Furthermore, commercial initiatives and manufacturers such as a5Systems, 6HARMONICS, Airband, and First Broadband provide technological services for rural broadband connectivity, the industrial Internet of Things (IIoT), communications for critical infrastructure and environmental monitoring, among other applications, all operating over TVWS. Table 2.5 shows the main parameters of the secondary networks implemented according to the 802.22 and 802.11af technologies. It is expected for a secondary network to be used for similar purposes as a typical WiFi network, being Internet access the most prominent application.

	802.11af	802.22
Coverage	Indoor: up to 100 m	17 - 33 km
-	Outdoor: up to few km	
Max delay spread	Indoor: less than 1 $\mu sec$	11 - 60 $\mu sec$
	Outdoor: 1 - 10 $\mu sec$	
Total bandwidth	5, 10, 20, 40 MHz	6, 7, 8 MHz
Maximum data rate	$12 \mathrm{~Mb/s}$	$22{,}69~{\rm Mb/s}$
Payload modulation	BPSK, QPSK	QPSK
	16-QAM, 64-QAM	16-QAM, 64-QAM

Table 2.5: Comparison parameters of 802.11af and 802.22 standards

#### Coexistence

Coexistence between cognitive radio networks both fixed and mobile is an active topic under study. Along that line, the IEEE 802.19.1 [22] standard defines a coexistence framework for cognitive networks over TVWS, but it is not yet clear where to implement the proposed entities, such as the coexistence manager (CM), the coexistence discovery and information server (CDIS), and the coexistence enablers (CE), to guarantee that any secondary network will be discovered and managed [101]. Although the coexistence framework may provide the operating parameters (e.g., the frequency, the transmission power, the modulation, among others) that avoid interference among networks, changing the operating parameters in each network is optional to date, which means that, in terms of the aforementioned standard, the coexistence between the networks has no warranties. Furthermore, the mobility in the cognitive vehicular networks is a condition that should be included to propose efficient coexistence methods, considering that the residence time within the secondary network coverage area is variable, and the coexistence process must be low latency to guarantee the delay requirements of applications [12].

Coexistence between cognitive vehicular networks and 802.22 networks has been explored by Han *et al.* [49], considering that 802.22 networks use time division multiple access (TDMA) to share the channel. The authors propose a coexistence 802.22-CVN framework where the upstream scheduling information of 802.22 Consumer Premise Equipment (CPE) devices is included in the downstream messages that are periodically broadcast by the 802.22 base station (BS). The CPE and the vehicles obtain the 802.22 upstream scheduling information by listening to the downstream messages from the BS, and the vehicle can access the TVWS channel according to the upstream scheduling information. They propose that the low-power CVN and 802.22 networks could reuse the same TVWS using adaptive resource allocation, thus avoiding mutual interference and reducing the need to search for another TVWS for the CVN when there are secondary networks in the area. Their resource allocation mechanism seeks to maximize the transmission rate in the vehicular network, penalizing the violation of restrictions such as the maximum transmission power allowed for secondary users and the interference caused to the secondary network.

### 2.6 Discussion

Most of the reviewed articles that study the feasibility of using TVWS for vehicular communications only consider the presence of the primary TV users. On the one hand, previous works propose a geolocation database that stores the information about the TV channel allocation, which is mainly used for V2I scenarios because the vehicles must request the available TVWS before transmission. On the other hand, for V2V scenarios, the reviewed articles propose to use a spectrum sensing technique to determine if a TV transmitter occupies a TV channel; the decision can be made using cooperative sensing among vehicles to improve the reliability. Once a TVWS is detected, the reviewed articles assume that the TVWS remains available along the road, focusing the attention to other aspects of the interworking. However, the presence of active IEEE 802.22 and IEEE 802.11af networks may affect the use of TVWS, and the coexistence between secondary networks and vehicular communications has been little studied.

Furthermore, coexistence among secondary networks has been studied considering that the opportunistic spectrum is used for different scenarios. Coexistence methods among secondary networks over TVWS are defined in the IEEE 802.19.1 standard [22], where a coexistence framework based on entities such as managers, enablers, and information servers interacts to collect operational settings and adjust the high-level operating parameters such as the operating channel and the power transmission of the secondary networks to avoid mutual interference. Also, each network has to agree on allowing automatic setting reconfiguration to operate under specific conditions to avoid the mutual interference between networks.

Although the IEEE standard 802.19.1 proposes different methods and algorithms for coexistence between secondary cognitive networks, the complexity of the architecture and the non-obligatory adjustment of the operating parameters to achieve coexistence avoiding mutual interference have made implementing the standard impractical. On the other hand, [49] proposes a mechanism for adjusting operational parameters only in the vehicular network to maximize throughput, penalizing non-compliance with the maximum power constraint and the interference caused to the fixed network. However, this mechanism does not guarantee the technical protection of the fixed secondary network operation beyond reducing throughput in the vehicular network as the incentive to choose the operating parameters in the vehicular network.

## Chapter 3

## Coexistence of Cognitive Networks for Vehicular Communications on TV White Spaces

This chapter presents the coexistence scenario for vehicular communications on TVWS Spectrum. First, the general scenario of study for vehicular networking on TVWS spectrum, also known as vehicular dynamic spectrum access (VDSA), is presented. The scenario includes the TV bands primary users, secondary networks, and vehicular networks. Then, a coexistence scenario among fixed secondary and vehicular networks based on the IEEE standard 802.22 is presented. Finally, the White Space Resource Sharing mechanism is detailed.

### 3.1 Introduction

Channel availability is a fundamental requirement for the successful operation of vehicular applications, but the lack of available spectrum for wireless technologies is a threat for current and future applications designed for ITS. For example, in the presence of traffic congestion, the sending of a critical warning message may fail when concurrent transmissions overflow the control channel assigned to vehicular communications. Therefore, the dynamic and opportunistic use of available spectrum, coupled with cognitive radio technology, has been proposed to ensure more available channels and successful transmissions in both fixed and mobile environments.

Dynamic Spectrum Access is the opportunistic detection and usage of the available radio spectrum. DSA for vehicular communication (VDSA) has been explored as an alternative or complementary option to dedicated short-range wireless communications (DSRC) [72], the technology for vehicular communications in the 5.9 GHz band. VDSA considers the use of available channels in TV frequency bands (470 MHz-806 MHz), known as TV White Spaces (TVWS), to exploit the propagation features of frequencies below 1 GHz, achieving higher transmission distances in vehicle-to-everything (V2X) scenarios. Among other reasons, TVWS may exist in a specific area when there are no TV transmitters using the channel in the area of interest or when a TV transmitter uses the channel in the area, but the TV transmission is OFF during specific hours (e.g., at night).

Spectrum regulators have defined the technical requirements for allowing DSA in TV bands to guarantee the protection of the primary user against harmful interference [70, 71, 102]. Secondary users interested in the TVWS have to make sure the channel is available before transmission and be ready to release the channel when a primary user requires it. Accordingly, IEEE has published two related standards for wireless communications for TVWS. The IEEE 802.11af standard is the specification of physical and medium access control layers for wireless local area networks employing TV bands [99], whereas the IEEE 802.22 standard specifies physical and medium access control layers for cognitive radio networks over TV bands [23] (see Section 2.4 for more details of the standard).

The traditional approach to VDSA has been to exploit TVWS as an offloading channel when traffic congestion causes the degradation of the DSRC control and data channels [46], and to provide an exclusive channel for Internet access [103]. Results have shown that the use of TVWS may increase the dissemination distance for safety messages in vehicular adhoc networks (VANETs), reducing collisions and latency caused by multi-hop dissemination [104, 93]. The work that has been done to date on VDSA on the TVWS spectrum reflects the fact that the TV channel is only occupied by a TV transmitter, and opportunistic access is available outside of the transmitter's coverage area [46, 44, 45, 47, 79, 86]. In these scenarios, vehicles travel across a route where the availability of TV channels depends on the distance between the TV transmitter and the vehicles. Channel availability is checked accessing a geolocation database with information of the spectrum occupation [46, 44, 45] or using a spectrum detection technique such as energy detection, cyclostationary analysis, or matched filter detection [47, 79, 86, 87].

A more realistic VDSA scenario should take into consideration the presence of other secondary opportunistic networks when they operate on the TVWS spectrum, potentially affecting the vehicular communications when the vehicles meet with these secondary networks on the route. Since the TVWS spectrum provides resources for secondary spectrum usage, all the secondary devices have the same right to use the TVWS channel. However, the impact of other opportunistic networks on the dynamic spectrum access for vehicular networking over TVWS is still under exploration.

### 3.2 Modeling primary and secondary TV bands users for vehicular networking on TVWS

Figure 3.1 illustrates the general scenario of study, where there is an active TV transmitter providing TV services using, for example, channel 52. Outside the coverage area of the TV transmitter, there is a V2V link established over the same channel because it has become a white space in that area. However, there is an IEEE 802.11af fixed secondary network (also known as a White-Fi network) farther along on the route, which is also using channel 52. The following presents a more detailed model of the TV primary users, the White-Fi network, and the vehicular network intending to access the TV channel.



Figure 3.1: Scenario of study: opportunistic access for vehicular networking considering the presence of primary TV users and secondary White-Fi networks over the TV band. Source: Own elaboration.

#### 3.2.1 TV primary users

Primary users of TV bands are the analog and digital TV transmitters, which are located in valleys or low to medium-sized mountains, providing directional coverage toward densely populated areas. When there is an active TV transmitter in the area, the channel assigned to the transmitter remains busy within the TV transmitter's coverage area, which means the channel cannot be used for secondary users inside that coverage area. However, spectrum regulators define that a TVWS may be used if the received power of the TV signal is lower than a specific threshold; in the case of the FCC, the detection threshold is -114 dBm [70].

In the presence of an active TV transmitter, opportunities for VDSA must be located outside the TV transmitter's coverage area. One way to estimate the TV transmitter's coverage area is by calculating the maximum distance between the transmitter and the potential receivers, in which the received power is higher or equal to the detection threshold. The received power is calculated using a long scale path loss model since the frequencies of TV service (470-806 MHz) have a high obstacle penetration. In [105], several path loss models to predict TV service coverage for secondary use are compared with path loss measurements in urban, suburban, and rural environments. Among the evaluated models, Hata [106] and Hata-Davidson [107] models showed similar results regarding path loss. The Hata model is used for frequency ranges of 100 MHz to 1,500 MHz, distances of 1 km to 20 km, base station antenna heights between 30 m and 200 m, and reception antenna heights between 1 m and 10 m.

Nevertheless, the TV transmitter's coverage is expected to be larger than 20 km in some rural areas. Therefore, the Hata-Davidson model is adopted in the system, which incorporates a correction factor to increase the distance between the transmitter and the receiver up to 300 km. The received power  $P_{Rx}$  corresponds to:

$$P_{Rx}(\mathrm{d}B) = P_{Tx} - PL_{HD},\tag{3.1}$$

where  $P_{Tx}$  is the transmitted power and  $PL_{HD}$  is the Hata-Davidson path loss, which is describe as follows.

#### Hata-Davidson Pathloss Model

The Hata-Davidson path loss model includes correction factors to the well-known Okumura-Hata path loss model in order to expand the model's parameters [107]. The Hata-Davidson pathloss  $PL_{HD}$  is calculated as follows.

$$PL_{HD}(dB) = PL_{Hata} + K_{Davidson}, \qquad (3.2)$$

where  $PL_{Hata}$  is the Hata path loss and  $K_{Davidson}$  is the Davidson correction factor.  $K_{Davidson}$  includes several terms to adjust the loss due to the expanded parameters:  $A(h_{bs}, d)$  and  $S_1(d)$  are distance correction factors;  $S_2(h_{bs}, d)$  is the base station antenna height correction factor;  $S_3(f)$  and  $S_3(f, d)$  are the frequency correction factors. Table 3.1 depicts the main variables used for the  $K_{Davidson}$  derivation.

Variable	Definition	
$PL_{HD}$	Hata-Davidson path loss (dB)	
$PL_{Hata}$	Hata path loss (dB)	
$K_{Davidson}$	Davidson correction factor (dB)	
$PL_U$	Hata path loss in urban area (dB)	
$PL_S$	Hata path loss in suburban area (dB)	
$PL_R$	Hata path loss in rural area (dB)	
d	Distance between transmitter and receiver (km)	
f	Transmitter frequency, $150 \le f \le 1000(MHz)$	
$h_{bs}$	Transmitter antenna height (m)	
$h_m$	Receiver antenna height (m)	
$A(h_{bs}, \mathbf{d})$	Distance correction factor	
$S_1(\mathbf{d})$	Distance correction factor	
$S_2(h_{bs}, \mathbf{d})$	Base station antenna height correction factor	
$S_3(f)$	Frequency correction factor	
$S_4(f, \mathbf{d})$	Frequency correction factor	

Table 3.1: Summary of variables in Hata path loss model and  $K_{Davidson}$  derivation.

The equations to calculate the correction factors are given below.

$$K_{Davidson} = A(h_{bs}, d) - S_1(d) - S_2(h_{bs}, d) - S_3(f) - S_4(f, d)$$
(3.3)

$$A(h_{bs}, \mathbf{d}) = \begin{cases} 0, \mathbf{d} < 20km \\ 0.62317(\mathbf{d} - 20)[0.5 + 0.15\frac{h_{bs}}{121.95}], 20km \le \mathbf{d} < 300km \end{cases}$$
(3.4)

$$S_{1}(d) = \begin{cases} 0, d < 64.38km \\ 0.174(d - 64.38), 64.38km \le d < 300km \end{cases}$$
(3.5)

$$S_2(h_{bs}, \mathbf{d}) = 0.00784 \left| log_{10}(\frac{9.98}{\mathbf{d}}) \right| (h_{bs} - 300), h_{bs} < 300m$$
(3.6)

$$S_3(f) = \frac{f}{250 \log_{10}(\frac{1500}{f})} \tag{3.7}$$

$$S_4(f, d) = \left[0.112 \log_{10}(\frac{1500}{f})\right] (d - 64.38), d > 64.38km$$
(3.8)

According to the path loss calculated in Eq. (3.2) and the detection threshold established by the FCC, the coverage areas of a typical TV transmitter in urban, suburban, and rural areas are estimated. Figure 3.2 shows the received power for distances up to 100 km between the TV transmitter and receiver. Table 4.6 lists the parameters for transmission and reception of the primary users. Results show that a channel assigned to an active TV transmitter remains busy for reception distances up to 75 km, 125 km, and 220 km in urban, suburban, and rural environments, respectively. For greater distances, the channel can be considered as TVWS and could be used opportunistically. However, the vehicle should use the energy detection VDSA technique to sense the channel occupation. As mentioned in Section 2.3, energy detection is the predominant spectrum detection technique in V2V scenarios because of its low latency execution. Since the FCC determines that a portable secondary device must verify the TV occupation every 100 m to protect TV primary users against harmful interference, a spectrum detection technique with low-latency is more convenient for vehicular secondary users that travel at 30-120 km/h. Note that Figure 3.2 shows a jump of the received power between kilometers 19 and 20; this is because the distance correction factor,  $A(h_{bs}, d)$ , employed in the Hata-Davidson model is a piece-wise function defined for the intervals [1, 20) km and [20, 300) km.

#### 3.2.2 White-Fi network description and traffic modeling

Considering the deployment of TVWS secondary networks (i.e., 802.11af or IEEE 802.22 networks), the main interest is to analyze if there are possible opportunities for vehicular communications even in the presence of such secondary users. In this case, it is necessary to estimate the coverage area of the White-Fi access point (AP) to define a bounded area where the TVWS could be busy. The scenario of study considers an AP antenna height of 30 m, which is the height of a 10-floor building in an urban area, or an antenna on a water tank or a small hill in a rural area. All of the receivers of the secondary users (i.e., White-Fi users or vehicular users) are assumed to have an antenna height of 1.5 m.

Fixed secondary users must limit the transmission power up to 36 dBm [70] according to the FCC. To estimate the AP coverage, the received power depending on the distance between the AP and the potential receivers according to the Hata-Davidson model is calculated. Since the White-Fi AP is also a secondary user, there is no specific detection threshold from the



Figure 3.2: Received power depending on the distance between a TV transmitter and a potential TV receiver.

transmissions of the secondary users to determine that a TVWS is available. Hence, the receiver sensitivity is employed to determine the area of coverage of the White-Fi network. According to the IEEE 802.22 standard, the receiver sensitivity for a successful reception is -90 dBm. Table 4.6 shows the parameters employed for transmission and reception of the White-Fi network system.

Figure 3.3 illustrates the received power when the distance between the AP and the potential receiver increases. According to the Hata-Davison model, the White-Fi network has a radio range of approximately 2 km in urban areas, 3.2 km in suburban areas, and 11 km in rural areas. Vehicular communications intending to use the TVWS inside such coverage areas should consider the channel occupation derived from the White-Fi network users' traffic.

#### White-Fi network traffic

A White-Fi network is expected to be used for similar purposes as a typical WiFi network with Internet access as the most prominent application. White-Fi could be also used to extend the coverage area of a monitoring network. The medium access channel in a White-Fi network is carrier sense multiple access (CSMA), and for this scenario of study the White-Fi network traffic is modeled in two ways. First, with sources of traffic that occupy the channel following a uniform distribution probability. Four channel occupation policies with 10%, 30%, 50%, and 70% probabilities of busy channel ( $P_{bc}$ ) were defined. The probability density function of the uniform distribution f(x) is:



Figure 3.3: Received power depending on the distance between the White-Fi access point and a potential TVWS receiver.

$$f(x) = \begin{cases} 0, x < a \\ \frac{1}{b-a}, a \le x \le b, \\ 0, x > b \end{cases}$$
(3.9)

where a and b are two boundaries that depend on the observation time, for example, from 0 to 100 ms. Figures 3.4a and 3.4b show examples of the 10% and 50% occupation policies. Second, with sources of bursty traffic that represent the Internet traffic. This traffic follows a Pareto ON/OFF distribution, where packets are sent during ON periods according to the mean burst time  $(b_t)$ , and no packets are sent during the OFF periods defined by the mean inter-arrival time ( $i_t$ ). Figure 3.4c shows an example of the Pareto occupation policy.

To establish the ON/OFF periods in the Pareto distribution, the packet duration  $p_{dur}$  and burst length  $b_l$  (in packets) are calculated according to:

$$p_{\mathrm dur} = \frac{p_s}{R},\tag{3.10}$$

$$b_l = \frac{b_t}{p_{\rm dur}},\tag{3.11}$$

where  $p_s$  is the packet size in bits, R is the data rate in bits per second during the burst, and  $b_t$  is the mean burst time. The burst length  $b_l$  and the idle time  $i_t$  correspond to the expected value  $E(X) = b_l$  and  $E(Y) = i_t$  of the Pareto distribution for each variable:



(c) Pareto distribution with  $b_t = 10 \text{ ms and } i_t = 20 \text{ ms.}$ 

Figure 3.4: Examples of channel occupation policies: (a, b) uniformly distributed probability of busy channel and (c) burst traffic. Source: Own elaboration.

$$E(X) = b_l = b_1 * \frac{a}{(a-1)},$$
(3.12)

$$E(Y) = \mathbf{i}_t = b_2 * \frac{a}{(a-1)},$$
(3.13)

where a > 1 is the Pareto shape parameter. The Pareto scalar parameters  $b_1$  and  $b_2$  are extracted from E(X) and E(Y) as follows:

$$b_1 = b_l * \frac{(a-1)}{a},\tag{3.14}$$

$$b_2 = i_t * \frac{(a-1)}{a}.$$
 (3.15)

Three configurations of the Pareto distribution were defined to model the bursty traffic according to values previously reported in the literature [108], [109]:  $b_t = 10$  ms and  $i_t = 20$  ms;  $b_t = 20$  ms and  $i_t = 10$  ms; and  $b_t = 20$  ms and  $i_t = 20$  ms.

#### 3.2.3 Vehicular network modeling

In the scenario of interest, the intend is to analyze the impact of the vehicles' speed on the perceived spectrum availability for vehicular dynamic spectrum access. The FCC determines that portable/mobile secondary devices in TV bands must verify the channel occupation every 100 m, to detect the possible presence of a primary user. In this study, the verification distance is also used to establish the presence of other users within the White-Fi network. Considering that the time it takes for a vehicle to travel 100 m depends on its speed, two



Figure 3.5: Differences in channel verification times for two vehicles traveling at 30 km/h and 90 km/h to cover a distance of 400 m. The channel occupation policy follows a Pareto Distribution. Source: Own elaboration.

vehicles with different velocities will meet the verification time (i.e., will cover the 100 m) at different moments. This situation causes the vehicles to have different perceptions of channel occupation, which is illustrated in Figure 3.5. For example, two vehicles moving at 30 km/h and 90 km/h, respectively, and both traveling 400 m, detect different channel occupations as follows: the vehicle moving at 30 km/h travels 100 m in 12 seconds, whereas the vehicle moving at 90 km/h travels 100 m in four seconds. If the channel occupation along a route is represented on a timeline, the opportunities for VDSA could be different for each vehicle as shown in Figure 3.5.

The scenario of study described above presents the most relevant elements to enable vehicular communications in the TVWS spectrum. The secondary network was modeled according to the IEEE standard 802.11af where the medium access channel is CSMA, then the channel usage is random access following one of the channel occupation policies. This scenario is used to analyze the impact of vehicle speed and channel verification distance on the perception of channel occupancy and the identification of transmission opportunities on the TV band. Evaluation and simulation results are described in Section 4.1.

The other IEEE standard for secondary spectrum access on the TV bands is the IEEE 802.22 [23], which also has the potential to provide connectivity to Intelligent Transportation Systems with cost-effective, extended coverage [98]. The following section describes the scenario of study focused on the coexistence between fixed secondary and vehicular networks, analyzing the TVWS channel sharing between the overlapping fixed secondary and vehicular networks, and presenting the proposed resource sharing mechanism. For the coexistence scenario, it is assumed that secondary fixed and vehicular networks only operate on vacated TV channels, with no primary users. The georeferenced database provides the list of vacated TV channels. Therefore, the TV band primary users are omitted in the coexistence scenario to focus on the coexistence between the vehicular and secondary networks.

### 3.3 Coexistence between fixed and mobile (vehicular) secondary groups on the IEEE standard 802.22

Fig. 3.6 illustrates an example of coexistence between fixed and mobile secondary groups in an A-WRAN as defined in the IEEE 802.22 standard [23]. Fixed or mobile groups operate as networks; therefore, the term *network* is used to refer to a group. In the scenario, there are five fixed secondary networks (FSN) (FSN<sub>1</sub>, ..., FSN<sub>5</sub>) managed by the A-BS using three TVWS channels (CH<sub>1</sub>, CH<sub>2</sub>, CH<sub>3</sub>). FSNs provide different services such as broadband Internet access for rural communities, connectivity for IoT deployments, and emergency network deployments. According to [23], several FSNs could use the same TVWS channel when networks are in non-overlapping areas, for example, FSN<sub>4</sub> and FSN<sub>5</sub> are non-overlapped so both networks use TVWS CH<sub>3</sub>.



Figure 3.6: Example of advanced fixed and vehicular secondary networks managed by a single A-BS. Source: Own elaboration.

Each FSN has one R-CPE as the central node and may have one or several A-CPEs connected. In this example, the A-BS and all the R-CPEs are connected through CH<sub>1</sub> to receive configuration parameters and send uplink traffic generated by A-CPEs. The R-CPE manages the A-CPEs' access to the allocated channel in each FSN, following a frame structure with upstream and downstream sub-frames. The frame structure of the 802.22 system is dynamically partitioned according to Time Division Duplex (TDD). The downstream medium access is time division multiplexed (TDM) for transmissions from the A-BS to A-CPEs/R-CPEs. The upstream medium access is managed using Demand Assigned Multiple Access (DAMA)/OFDMA at the A-CPEs/R-CPEs for coordinating the channel access. The downstream sub-frame includes the information for the upstream sub-frame assignment (i.e., burst start time, burst length, and sub-channel usage) to be processed by all the A-CPEs/R-CPEs within the A-BS coverage area.

In the example of Fig. 3.6, there are several vehicular networks  $(VN_1, VN_2, and VN_3)$  arriving to the coverage area of the A-BS. The standard defines the group resource allocation on PHY-OM2 to manage the spectrum sharing between fixed and mobile secondary groups within the A-BS coverage area. The A-BS creates a new group, identifies the devices that belong to the group, and allocates the resources on a group basis. The allocated TVWS channel (named backup channel in [23]) and candidate-for-use TVWS channels are selected according to the mobility and location of the R-CPEs.

The group resource allocation of the standard specifies that several fixed groups may use the same channel unless the R-CPEs' coverage ranges overlap (e.g., in Fig. 2.3, Group 1 and Group 2 are overlapped). For fixed groups overlapping, the A-BS allocates a different channel to each group to avoid interference among groups. The same applies for overlapping fixed and mobile groups, as well as for overlapping mobile groups, if there are remaining TVWS channels available. The case of a fixed and a vehicular (mobile) groups overlapping is illustrated in Fig. 3.7. If there are no TVWS channels available, or when groups share the TVWS channel for spectrum efficiency purposes, the standard states that "the A-BS may use a mechanism for proper resource sharing between the overlapping groups or the A-BS and some groups after channel switching. [23]". However, such a resource sharing mechanism is not defined in the standard.



Figure 3.7: Scenario of study. Source: Own elaboration.

In the following section, the proposed WSRS mechanism is introduced, which allows the coexistence of fixed and vehicular secondary networks that share the TVWS channel when

operating in overlapping areas. Our proposal is to use a resource-sharing approach based on joint power control and channel allocation to make it feasible for both networks to share the same TVWS channel. Below, the study scenario is described.

## 3.4 White Space Resource Sharing (WSRS) Mechanism

Each vehicle network (VN) acts as a cluster of N vehicles where one vehicle is an R-CPE, which will be known as the Manager Vehicle (MV) in charge of resource allocation (detailed later in Section 3.4.2). The remaining N-1 vehicles are A-CPE. The MV communicates to the vehicular network with the A-BS to send the uplink traffic of the vehicular network.

The White Space Resource Sharing (WSRS) mechanism for the coexistence between FSNs and VN involves scheduling vehicular transmissions depending on the FSN upstream scheduling information. Vehicles and fixed nodes may transmit at the same time using an adjusted transmission power, reducing mutual interference. The WSRS mechanism consists of three main functionalities shown in Fig. 3.8:



Figure 3.8: White Space Resource Sharing (WSRS) Mechanism. Source: Own elaboration.

- Spectrum Usage Learning Function (SULF): Learn the TVWS channel occupation due to the FSN traffic by listening to the FSN downstream.
- Resource Allocation Function (RAF): Execute the power control and channel access processes to generate the scheduling information for the vehicular network.
- Scheduling Information Updating Function (SIUF): Update the scheduling information (i.e., the transmit power and the slot-time) to the vehicles via broadcasting.

#### 3.4.1 System's operation

The coexistence among FSNs and VSNs is based on the channel access shown in Fig. 3.9. A TVWS channel is divided in sub-channels, and each sub-channel is divided in upstream

and downstream. Upstream and downstream access in 802.22 systems is TDM-based. The frame structure is partitioned into upstream and downstream sub-frames using TDD. Bursts conform the upstream sub-frame; each burst is a two-dimensional segment of OFDM sub-carriers (frequency-domain) and symbols (time-domain). Upstream bursts are allocated by contention or polling, depending on the network's configuration; then, the A-BS (or the R-CPE acting as the relay node between the A-CPEs and the A-BS in the network) includes the upstream scheduling information in the downstream sub-frame and broadcasts it.



Figure 3.9: Upstream sub-frame scheduling for vehicular and fixed network. Source: Own elaboration.

It is assumed that the TVWS channel allocated to each FSN is divided into sub-channels equal to the number of fixed users. Each fixed A-CPE uses an exclusive frequency range within the TVWS channel. The number of sub-channels and the bandwidth may be adjusted up to 105 sub-channels according to the IEEE 802.22 standard. When the number of fixed A-CPEs increases, the channel access in the time domain may also be adjusted to extend the network capacity, making it possible for several fixed A-CPEs to share the same sub-channel at different times.

#### 3.4.2 Resource allocation in WSRS

Using the example scenario depicted in Fig. 3.7 as a reference, in this case, the  $R-CPE_5$  manages the channel access for the fixed network FSN<sub>5</sub>. When the vehicular network VN<sub>3</sub> enters into the coverage area of  $R-CPE_5$ , the MV<sub>3</sub> receives the upstream scheduling information of  $R-CPE_5$ , from which it learns the TVWS channel utilization and the location of

fixed A-CPEs. With this knowledge, the  $MV_3$  executes the resource allocation based on power control and channel allocation to generate the vehicular transmission scheduling for each time frame (see Fig. 3.9). When the vehicles are informed of the WSRS scheduling, concurrent transmissions are placed from A-CPEs and VNs over the same TVWS channel.

Allocation based on the power control process and the channel control process is detailed below. Table 3.2 summarizes the most important symbols and variables.

Symbol	Variable
N	Number of vehicles
M	Number of fixed nodes
$V_p$	Vehicles' position
$C_p$	A-CPEs' location
$P_{A-CPE}$	A-CPEs' transmit power
f	Operational frequency (MHZ)
$c_j$	A-CPE ID j
vi	Vehicle ID i
j	A-CPE transmitting during burst $j$
$v_{i}^{T}$	Transmitter vehicle in i-link
$v_{i}^{R}$	Receiver vehicle in i-link
$d_v$	Inter-vehicular distance (m)
$p_{i}^{j}$	Vehicle's i transmit power on burst $j$
$p_c^j$	A-CPE's transmit power on burst $j$
$L_{v_i}$	Pathloss in i-th vehicular link
$L_j^{i}$	Pathloss in link between $c_j \ge v_i^T$
$W_j$	Bandwidth of burst $j$
$I_{\rm i}^j$	Interference over $v_i^R$ caused by <i>j</i> -th S-CPE
$I_j^{\mathrm{i}}$	Interference over <i>j</i> -th A-CPE caused by $v_i^T$
Ni	Noise over $v_i^R$
Xi	Channel capacity for vehicular link i (Mb/s)
J(X)	Jain's Index
$S_R$	Service Rate
E	Spectral Efficiency
$SINR_V$	Signal-to-interference plus noise ratio of vehicular network

Table 3.2: WSRS mechanism symbols and variables

#### Power Control Process (PCP)

The power control process (PCP) calculates the required transmission power for each link between two vehicles (e.g., vehicle  $v_i$  and  $v_{i+1}$ ), depending on the path loss of the vehicular link and the received interference caused by the fixed network. In the study scenario, the N vehicles are connected through N-1 one-to-one links between the transmit vehicle  $v_i^T$  and the receiver vehicle  $v_i^R$ .

For path loss in the vehicular links  $L_{v_i}$ , a path loss model suitable for vehicular networks operating at TV band frequencies [110] is used. This path loss model is based on several channel measurements for different environments (i.e., urban, suburban, rural, and highway), road traffic densities, line-of-sight (LOS) and non-line-of-sight (NLOS) paths, and vehicle speeds. The propagation model parameters are derived from the channel measurements and used in the classical log-distance power law to calculate the path loss. The required parameter for this path loss model is the vehicular positions  $V_p$  to obtain the inter-vehicular distance  $d_v$ .

Interference made by the fixed nodes is the other aspect that affects the required transmit power to maintain the vehicular links within the FSN coverage area. This interference  $I_i^j$ received by the vehicle  $v_i^R$  from the A-CPE *j* signal is calculated as

$$I_{i}^{j} = p_{c}^{j} - L_{j}^{i}, \qquad (3.16)$$

where  $p_c^j$  is the transmit power of the A-CPE j and  $L_j^i$  is the path loss between the A-CPE j and the vehicle  $v_i^R$ , as shown in Fig. 3.10.



Figure 3.10: Power control function variables. Source: Own elaboration.

The PCP defines a signal-to-noise plus interference ratio to maintain the vehicular links  $(SINR_V)$  within the FSN coverage area, considering the path loss of the vehicular link and the interference caused by the fixed nodes. SINR in terms of transmit power and path loss is calculated as follows:

$$SINR_{V} = \frac{\frac{p_{i}^{j}}{L_{v_{i}}}}{\frac{p_{c}^{j}}{L_{j}^{i}} + W_{j}N_{0}}$$
(3.17)

The required transmit power for each vehicular link i to transmit at the same time as the

$$p_{i}^{j} = SINR_{V} + L_{v_{i}} + p_{c}^{j} - L_{i}^{i} + W_{j}N_{0}, \qquad (3.18)$$

The output of PCP is an N-1 x M matrix P, which contains the required transmit power for all combinations of vehicular link  $(v_i - v_{i+1})$  transmitting at the same time as each A-CPE. If  $p_i^j$  is more than the 20 dBm maximum transmit power allowed for secondary mobile users according to the IEEE Std. 802.22, the required power is not feasible. To simplify the notation in what follows, the vehicular link  $(v_i - v_{i+1})$  will be identified as i.

#### Channel Allocation Process (CAP)

The channel allocation process (CAP) generates a set of triads  $K = \langle i, c_j, p_i^j \rangle$  where the vehicular link i uses the transmit power  $p_i^j$  to transmit in the same burst to the A-CPE  $c_j$ . If there is no feasible power for the vehicular link i, its triad will be  $\langle i, 0, 0 \rangle$ . The inputs of CAP are the matrix P as well as the matrix D that contains the distance between all the A-CPEs and the vehicles. CAP seeks to match the vehicular link i with the farthest A-CPE j as a strategy to protect the A-CPEs' transmission from the vehicular link interference.

To maximize the distance between vehicular links and A-CPEs, CAP applies the Hungarian algorithm [111] to matrix D. The combinations i,  $c_j$  selected by the Hungarian algorithm correspond to the vehicular link i and A-CPE j pair that may transmit at the same time. The corresponding  $p_i^j$  value is chosen from matrix P using the i, j index to form the triad  $\langle i, c_j, p_i^j \rangle$ .

#### Algorithm 1 White Space Resource Sharing (WSRS) Mechanism

**Input:**  $N, M, SINR_V, V_p, C_p, P_{A-CPE}$  and  $W_j, \forall j$ Output:  $K = \langle i, j, p_i^j \rangle$ 1: for Time frame do Calculate  $L_{v_i}, \forall v_i^T$ 2: Calculate  $L_j^i, \forall v_i^T, c_j$ 3: Calculate  $I_{i}^{j}, \forall v_{i}^{R}, c_{j}$  using Equation (3.16) 4: Calculate matrix  $P = \{p_i^j, \forall i, j\}$  using Equation (3.18) 5:Calculate matrix  $D = \text{distance}(v_i^T, c_i), \forall i, j$ 6:Solve Hungarian Algorithm for matrix D 7: $\mathbf{return} < \mathbf{i}, c_j, p_{\mathbf{i}}^j > \mathbf{\check{\forall}} v_{\mathbf{i}}^T, c_j$ 8: 9: end for

The proposed WSRS mechanism is an approach to improve network coexistence within the framework of the IEEE standard 802.22 by adjusting the operating parameters of the vehicular networks rather than those of the fixed networks. Section 4.3 presents the evaluation of the WSRS mechanism exploring the use case when vehicular networks (i.e., mobile secondary groups) move within an area where there are several fixed secondary networks (i.e., fixed groups), and there are not enough vacated TVWS channels to allocate, so overlapped fixed and vehicular networks must share the TVWS channel.

## Chapter 4

## **Evaluation and Results**

This chapter presents the evaluation of the scenarios to study the coexistence for vehicular communications in the TVWS spectrum:

- The feasibility study of TVWS spectrum sharing for vehicular communications in the general scenario considering primary TV band users and secondary networks. A new metric is introduced, the Channel Availability for Opportunistic Vehicular Access (CAFOVA), which relates the channel occupancy of the White-Fi network, the speed of the vehicle, and the channel verification distance. This study analyzes the impact of vehicle speed and channel verification distance on the perception of channel occupancy and the identification of transmission opportunities over the TV band.
- A comparative evaluation of two strategies for opportunistic vehicular access over the TVWS spectrum in the presence of other opportunistic networks is presented. One strategy is searching for a new TVWS channel for vehicular access when the current TVWS channel is occupied with other opportunistic secondary network. In the other strategy the vehicular networks share the TVWS channel with a fixed secondary. The comparative evaluation analyzes the channel access delay and the packet loss rate for both strategies.
- The performance evaluation of the White Space Resource Sharing mechanism for the coexistence among fixed secondary and vehicular networks according to the IEEE standard 802.22. This is an analysis of the impact of TVWS channel sharing between a vehicular network and secondary fixed networks in terms of the channel capacity offered to the vehicular network and attention to fairness and spectral efficiency of resource allocation.

## 4.1 Feasibility study of TVWS spectrum sharing for vehicular communications on TVWS spectrum

#### 4.1.1 Evaluation scenario

The evaluation scenario for this study is presented in Figure 4.1, where there is a TV transmitter offering the TV service, a White-Fi network outside the TV transmitter coverage



Figure 4.1: Evaluation scenario. Source: Own elaboration.

area, and a V2V link over the same TVWS coverage area as the White-Fi network. Table 4.1 presents the parameters employed in the scenario to model the primary TV service, the White-Fi network and traffic, and the vehicular network. The parameters follow the FCC requirements for TVWS [70], the IEEE 802.22 standard [100], and typical speed values employed in vehicular urban, suburban, and highway scenarios.

According to the White-Fi access point (AP) coverage range calculated in Section 3.2.2, the maximum White-Fi AP coverage is around twice the coverage radio (i.e., 4 km, 8 km, and 24 km, in urban, suburban, and rural areas, respectively). The channel occupation of the White-Fi network traffic is represented by a timeline where each millisecond is set to busy or idle according to the channel occupation policies described in Section 3.2.2. The vehicle moves along a straight route within the White-Fi coverage area. The evaluation considers several speed values to analyze the effect of the speed in the channel occupation perception. A vehicle senses the channel status every time it meets the channel verification distance; this is to ensure there is no TV primary user around, and also to determine if the TVWS is available or busy according to the channel occupation policy of the White-Fi network. The scenario is evaluated with several verification distances, including the 100 m specified by the FCC. The number of channel sensing verifications ( $C_{sv}$ ) depends on the length of the route ( $R_l$ ), and the channel verification distance (l), as follows:

$$C_{sv} = \frac{R_l}{l}.\tag{4.1}$$

To calculate the channel access opportunities, a new metric is introduced, known as *chan*nel availability for opportunistic vehicular access (CAFOVA) along the route, which represents the percentage derived from the number of times the TVWS is available  $(C_{sv}^A)$  with

Parameter	Value			
Primary users				
TV transmission power	70 dBm [112]			
TV transmitter antenna height	70 m			
Receiver antenna height	3 m			
Channel frequency	590 MHz			
Route length	100 km			
Detection threshold	-114 dBm [70]			
White-Fi ne	etwork			
AP power transmission	36 dBm [70]			
Channel frequency	590 MHz			
Height of the base station	30 m			
Height of the receiver	1.5 m			
Receiver sensitivity	-90 dBm [100]			
Busy channel probability	10%,  30%,  50%,  and  70%			
Mean burst duration	10  ms, 20  ms [109]			
(Pareto distribution)				
Mean burst inter-arrival time	10  ms, 20  ms [109]			
(Pareto distribution)				
Pareto shape parameter	1.5 [109]			
Vehicular network				
Vehicle speed	10 to 120 km/h $$			
Height of the receiver	1.5 m			
Receiver sensitivity	-90 dBm [100]			
Channel frequency	590 MHz			
Channel verification distance	25, 50, 100, and 200 m			

Table 4.1: Evaluation Parameters.

respect to the number of channel sensing verifications  $(C_{sv} = R_l/l)$  as follows:

$$CAFOVA = \frac{C_{sv}^A}{C_{sv}} \times 100\%.$$
(4.2)

CAFOVA considers the channel occupation policy of the White-Fi network, the speed of the vehicle, and the channel verification distance.

The evaluation scenario was built in Matlab and ran simulations to estimate the opportunities for VDSA across a route on urban, suburban, and rural areas. A Monte Carlo simulation was used to simulate the scenario 100 times for each combination of parameters, for a total of 28,000 simulations.

#### 4.1.2 Simulation Results

In the following, the discussion about the behavior of the CAFOVA metric under different channel occupation policies is presented. Also, the impact on the CAFOVA of the following variables is shown: the type of area of deployment, vehicle speed, and channel verification distances.

#### Impact of the type of deployment area

Results in Figure 4.2 show the CAFOVA metric within the White-Fi network coverage in a) urban small/large city, b) suburban area, and c) rural area. The channel occupation policy follows a uniform distribution whereas the verification distance l is set to 100 m, according to the FCC specifications. The CAFOVA obtained is around the complement of  $P_{bc}$  in all environments. Note also that the percentage of opportunities for vehicular access increases as the speed of the vehicle increases in the three areas under evaluation.

A statistical analysis was performed to understand the relationship between the CAFOVA,  $P_{bc}$  (10%, 30%, 50%, and 90%), the speed of the vehicle v (20, 50, 90, and 120 km/h), and the number of channel sensing verifications  $C_{sv}$  when l = 100 m (i.e., 41 verifications in urban scenarios, 84 verifications in suburban scenarios, and 241 verifications in rural scenarios). Table 4.2 shows the correlation analysis among the variables. The Pearson correlation value ( $P_c$ ) indicates the extent to which two variables are linearly related. The p-value ( $P_v$ ) determines if the correlation between the variables is statistically significant.

The analysis shows that the linear relationship between the CAFOVA and  $P_{bc}$  is -0.913, which means that the CAFOVA decreases when  $P_{bc}$  increases. Since the p-value is 0.00 and it is lower than the significant value 0.05, the correlation between these variables is statistically significant. Since the  $P_c$  between CAFOVA, v, and  $C_{sc}$  is 0.00, there is no linear relationship between these variables when they are analyzed together. The blank cells in Table 4.2 indicate the correlation and p-value are not calculated when it corresponds to the crossing of the same variable.

	CAFOVA		P	<b>)</b> bc	v	
	$P_c$	$P_v$	$P_c$	$P_v$	$P_c$	$P_v$
$P_{bc}$	-0.91	0.00			0.00	1.00
v	0.19	0.00	0.00	1.00		
$C_{sv}$	-0.17	0.00	0.00	1.00	0.00	1.00

Table 4.2: Correlation between the CAFOVA,  $P_{bc}$ , v, and  $C_{sv}$ .

In the analysis for bursty traffic, there are three different occupation policies: A ( $b_t = 20$  ms,  $i_t = 10$  ms), B ( $b_t = 20$  ms,  $i_t = 20$  ms), and C ( $b_t = 10$  ms,  $i_t = 20$  ms). A separate evaluation for urban, suburban, and rural areas is also provided. In this scenario, the vehicle is traveling at 50 km/h and the channel verification distance is 100 m. Results in Figure 4.3 show that the CAFOVA increases substantially when  $i_t$  is higher than  $b_t$ , regardless of the type of area. This is because the channel remains available more time between two data transmissions that are short. When  $b_t$  and  $i_t$  are the same, the CAVOFA is around 50% in all areas. The Pearson correlation and the p-value between CAFOVA,  $b_t$ ,  $i_t$ , and v are shown in Table 4.3. The correlation shows the positive linear relationship between the CAFOVA and the burst time.



Figure 4.2: CAFOVA in uniformly distributed channel occupations  $P_{bc}$ , when v is set to 20 km/h, 50 km/h, 90 km/h, and 120 km/h. l = 100 m.



Figure 4.3: CAFOVA in Pareto distributed channel occupations A, B, and C, when the area type is urban, suburban, and rural. v = 50 km/h, l = 100 m.

	CAFOVA		CAFOVA $b_t$		i <sub>t</sub>	
	$P_c$	$P_v$	$P_c$	$P_v$	$P_c$	$P_v$
$b_t$	-0.85	0.00			-0.50	0.00
i <sub>t</sub>	0.88	0.00	-0.50	0.00		
v	0.00	0.98	0.00	1.00	0.00	1.00

Table 4.3: Correlation between the CAFOVA,  $b_t$ ,  $i_t$ , and v.

#### Impact of the vehicles speed

Table 4.4 presents the statistical analysis based on Pearson correlation  $P_c$  and the p-value  $P_v$  when CAFOVA and v are analyzed for each  $P_{bc}$ , and for v between 10 to 120 km/h in the urban scenario. Since the correlation for both variables is 0.78 and the p-value is 0.00, there is a positive linear relationship between both variables. This means that the CAFOVA increases when the speed of the vehicle increases in scenarios where the White-Fi network traffic follows a uniformly distributed channel occupation  $P_{bc}$ .

Furthermore, Table 4.5 shows the Pearson correlation and the p-value between the CAFOVA and speeds ranging from 10 km/h to 120 km/h for the A, B, and C bursty traffic policies. Since the correlation is near to 0 and the p-value is higher than 0.05, there is no linear relationship between the CAFOVA and the speed of the vehicle when the White-Fi network traffic is bursty.

Table 4.4: Correlation between the CAFOVA and v for four different uniformly distributed traffic policies 10%, 30%, 50%, 70%. Urban scenario

	CAFOVA							
	$P_{bc} = 10\%$		$P_{bc} =$	30%	$P_{bc} = 50\%$		$P_{bc} = 70\%$	
	$P_c$	$P_v$	$P_c$	$P_v$	$P_c$	$P_v$	$P_c$	$P_v$
v	0.78	0.00	0.78	0.00	0.78	0.00	0.78	0.00

Table 4.5: Correlation between the CAFOVA and v for three different bursty traffic policies A, B, and C.

	CAFOVA					
	А		В		С	
	$P_c$	$P_v$	$P_c$	$P_v$	$P_c$	$P_v$
v	-0.001	0.99	-0.002	0.99	-0.007	0.97

#### Impact of the channel verification distance

To understand the effects of the channel verification distance, the CAFOVA is evaluated when the channel verification distance, l, differs from the one recommended by the FCC. Additional values of 25 m, 50 m, and 200 m are included in the evaluation. The number of channel verifications obviously depends on l and the route length. In the urban area where the route length is 4 km, the number of channel verifications is 161, 81, 41, and 21 times when l is 25, 50, 100, and 200 m, respectively. The vehicle is moving at an average speed of 50 km/h.

On the one hand, if the traffic of the White-Fi network follows a uniformly distributed  $P_{bc}$ , the CAFOVA is calculated to be around the complement of  $P_{bc}$ , as mentioned previously. Also, when the spacing distance between the number of verifications is longer, the CAFOVA is higher. Such results are illustrated in Figure 4.4. On the other hand, if the traffic of the White-Fi network follows a Pareto distribution, the behavior of the CAFOVA seems not to be sensitive to the channel verification distance l. Nevertheless, the CAFOVA clearly varies depending on the values of  $b_t$  and  $i_t$ . Such results are illustrated in Figure 4.5.

#### 4.1.3 Discussion

According to the results, it has been confirmed that, as expected, the nature of traffic in the White-Fi network impacts the channel availability for opportunistic vehicular access. This study analyzed seven channel occupation policies to model the traffic in the White-Fi network and considered several types of applications running on the network. On the one hand, given a channel occupied by bursty traffic, when the idle time is longer than the burst time, CAFOVA improves, i.e., there are more opportunities for the vehicular access because the channel remains idle more time. On the other hand, when the probability of busy channel follows a uniform distribution, the CAFOVA metric is around the complement of the probability of a busy channel.



Figure 4.4: CAFOVA in uniformly distributed channel occupations  $P_{bc}$ , when l is set to 25 m, 50 m, 100m, and 200 m. v = 50 km/h.

Speed has an impact on the CAFOVA when the channel occupation policy follows a uniformly distributed probability of a busy channel: the CAFOVA increases when the speed of the vehicle increases and the probability of a busy channel decreases. The reason is that the speed determines how long the vehicle will remain inside the White-Fi's coverage area. For example, a vehicle moving at 90 km/h observes a shorter timeline compared to a vehicle moving at 120 km/h. Although the probability of a busy channel is uniformly distributed, the number of busy milliseconds in a shorter timeline is proportional to the probability in the entire timeline. As a result, the percentage of opportunities for vehicular access is approximately the complement of the occupied channel probability, and increases linearly as the speed of the vehicle increases.

For bursty traffic, vehicle speed has little effect on the CAFOVA. Although vehicle speed determines when the channel occupation is sensed along the timeline, when the vehicle moves from 10 km/h to 120 km/h, there are no changes in the CAFOVA due to vehicles moving faster or slower inside the White-Fi's coverage area.

The channel verification distance is also analyzed, considering distances different from the 100 m recommended by the FCC. For the same trip, changing the verification distance affects the number and frequency of channel verifications. On the one hand, the results have confirmed that the CAFOVA increases when the channel verification distance increases, when the channel occupation policy follows a uniformly distributed probability of a busy channel. On the other hand, the CAFOVA is insensitive to changes in the number of verifications when the traffic follows a Pareto distribution.

According to the results, it is possible to enable opportunistic vehicular communications



Figure 4.5: CAFOVA in Pareto distributed channel occupations A, B, and C, when l is set to 25 m, 50 m, 100m, and 200 m. v = 50 km/h.

in the presence of White-Fi networks sharing the same TVWS. Although this study is not focused on determining how long the channel is available along the route, one feasible type of use, considering the channel opportunities observed, would be to use the TVWS as a backup technology for sending duplicate messages when the primary technology (e.g., DSRC channels) is highly congested. Sharing the same TVWS with a White-Fi network is feasible when the vehicles spend short times within the White-Fi coverage area, to avoid delays in searching for a new TVWS and to ensure vehicles tune to the same TVWS to maintain communications.

## 4.2 Evaluation of Opportunistic Access Strategies for Vehicular Networking Over TVWS spectrum

The feasibility study showed that both networks could coexist in the same TVWS channel, based on the CAFOVA metric that relates the channel occupation policy over the fixed cognitive network, the vehicle's speed, and the channel verification distance [14]. However, the coexistence between these networks would require a mechanism for channel access control to coordinate the TVWS channel usage. The other alternative is for the vehicular network to avoid using the occupied TVWS channel and starting the search for an available TVWS channel. However, the time spent looking for a new TVWS channel may degrade the application's performance if the vehicles need to verify a considerable number of TVWS channels before finding an available one. The following evaluation compares both alternatives to establish the channel access delay and the packet loss rate.

#### 4.2.1 Scenario of Study

Fig. 4.6 illustrates an example of the scenario of study, where there is an opportunistic vehicular network running an application based on clustering such as a routing protocol [113] over the TV channel 34, and there are several fixed secondary networks farther along on the route. The fixed secondary networks are operating on available TVWS channels via secondary base stations (BS); one or more of the secondary BS may use the TV channel 34 as long as their coverage areas do not overlap. Since the opportunistic access over TVWS spectrum provides the same right to access the spectrum for secondary users, both the vehicular network and the fixed secondary network have the same right to use the TVWS channel 34. This study assumes no coordination mechanism is employed between the opportunistic networks. When the vehicular cluster meets the fixed secondary network in channel 34, it should apply one of the strategies for opportunistic spectrum access. The opportunistic vehicular network crosses the coverage area of several secondary BS along the route. The details of such strategies are described as follows.



Figure 4.6: Scenario of study. Source: Own elaboration.

## Searching for a new TVWS channel for exclusive opportunistic vehicular access (strategy A)

This strategy requires the following steps before completion: first, the vehicles use a cooperative spectrum detection technique to decide if the current TVWS channel is available or busy [114]. If the TVWS channel is busy, the second step is the search for an available TVWS channel where the vehicles change the frequency to the next consecutive TVWS channel and execute the spectrum detection and decision step again. When the vehicles find an available TVWS channel, the third step is that vehicles move their data transmissions to the new TVWS channel. In case the selected TVWS channel becomes busy by a secondary network farther along the route, the vehicles shall rerun the process to find a new TVWS channel.

The channel access delay for this strategy is measured from the moment when the spectrum detection and decision process starts to the moment when the process finds an available TVWS channel. The channel access delay for strategy A  $(D_A)$  is calculated as follows:

$$D_A = N \times T_{SD},\tag{4.3}$$

where N is the number of TVWS channels that the vehicles have to change before they find an available TVWS channel, and  $T_{SD}$  is the time taken by the CR modules to perform the spectrum detection and decision process. Since there are several vehicles participating in the cluster-based application, there could be a basic mechanism for coordinating the search, which may consist of all the vehicles performing the cooperative sensing technique for searching the channel in order, i.e., jumping to the next consecutive TVWS channel and sensing if the channel is available or occupied. Other mechanisms are based on rendezvous protocols that help the vehicles find a common channel to establish communications [115]. In this work, the use of the basic mechanism is assumed.

#### Sharing the same TVWS channel with a fixed secondary network (strategy B)

In this strategy, when the opportunistic vehicular network meets a fixed secondary network in the same TVWS channel, the vehicles decide to stay and find an opportunity of transmission when the TVWS channel remains idle, similar to the carrier sense multiple access with collision avoidance (CSMA/CA) process. When the vehicles detect the channel is idle, they start a timer which is a delay time that vehicles must wait to ensure the channel is still idle; when the timer ends, the vehicles can use the TVWS channel to transmit the data. If the TVWS channel becomes busy before the timer ends, the vehicles must start to listen to the channel until it becomes idle again. The channel access delay for strategy B ( $D_B$ ) is calculated as follows:

$$D_B = T_L + T_D \tag{4.4}$$

where  $T_L$  is the time spent by the vehicles listening to the channel for the first time until it is detected that the TVWS channel is idle, and  $T_D$  is the timer that the vehicles must remain listening to ensure the channel is still idle.

The comparative evaluation is carried out using the NS-3 simulator, including the CR modules for spectrum detection and decision process. Both processes work jointly based on the Singular Value Decomposition (SVD) detection method, which is a cooperative energy detection process [116]. The vehicles execute the spectrum detection technique and report a detection bit to decide if the channel is busy or available. To obtain the results, the delay and packet loss rate is measured using the simulator developed by Palacios [117].

The evaluation scenario consists of a cluster of vehicles traveling across an area similar to the one depicted in Fig. 3.1, where there are 10 TVWS channels, and each TVWS channel has a fixed secondary network. Two probabilities of busy channel ( $P_B$ ) for the fixed secondary networks (i.e., 50% and 70%) as defined to study the average case and a near-to-worst case of channel occupancy.  $P_B$  means that the traffic in the TVWS channel follows a uniformly distributed probability of busy channel, then the TVWS channel will be occupied by the fixed secondary network, for example, 50% of the time during the spectrum detection and decision process. In strategy A, the cooperative spectrum detection and decision process takes  $T_{SD} = 0.5$  ms [100].

In strategy B, i.e., sharing the same TVWS channel with a fixed secondary network, the vehicles continuously listen to the channel, executing the spectrum detection and decision process as a background process. When vehicles detect a fixed secondary network, they stay in the same TVWS channel and start listening until the TVWS channel becomes idle. Then

the decision process starts the  $T_D$  timer (i.e.,  $T_D = 2 \times \text{Slot}$  Time in 802.11a [118]) to ensure the channel remains idle during that period, as a way to avoid a collision with the users of the fixed secondary network. When the  $T_D$  stops the vehicles can use the TVWS channel.

The packet loss rate  $P_{LR}$  is measured between two consecutive executions of the strategy using the received packet  $P_R$ , the lost packets  $P_L$  as follows:

$$P_{LR} = \frac{P_L}{P_L + P_R} \tag{4.5}$$

Using a Monte Carlo simulation, each strategy is simulated 10 times for each probability of busy channel for a total of 80 simulations. Table 4.6 shows the simulation parameters used in NS-3 simulator to set up the scenario. The propagation models provided by NS-3 to implement the vehicular communications, the TV transmitter and the secondary BS consider the effects of path loss, multipath, and shadowing [116]; the vehicular network moves with a constant velocity model.

Parameter	Value
Number of TVWS channels	10
Secondary BS for each TVWS channel	1
Secondary BS coverage area	3.2 km in urban areas[14]
Secondary BS Tx Power	10 kW [112]
TV Channel frequency	470 - 806 MHz [112]
TV Channel bandwidth	6 MHz [112]
Number of vehicles in cluster	5
Vehicle Tx Power	0.037 W [119]
Vehicle Rx Power	0.06 mW [119]
Speed of the vehicles	60  km/h
Mobility model	Constant Velocity Model
Probability of busy channel $(P_B)$	50% and $70%$
Sensing and detection time $(T_{SD})$	0.5 ms [100]
Timer for strategy B $(T_D)$	32 us [118]

 Table 4.6:
 Simulation
 Parameters

#### 4.2.2 Simulation results

Results in Fig. 4.7 show the channel access delay for the strategies depending on the probability of busy channel. When  $P_B=50\%$ , the 75% of the attempts to use the TVWS channel when the vehicular network shares the TVWS channel with the fixed secondary network (i.e., strategy B) were reached in maximum 100 ms; in strategy A, when the vehicles search for a new TVWS channel, the 75% of the attempts took up to 120 ms. Hence, strategy B reduces the average channel access delay by 15% when  $P_B = 50\%$  and a 16% when  $P_B=70\%$ , regarding strategy A. The minimum channel access delay is near to 20 ms for both strategies regardless  $P_B$ , and it is achieved when the current TVWS channel remains available, and the vehicles can continue to use it. The maximum channel access delay is different for each strategy and each  $P_B$ . In strategy A, the maximum delay should be expected when the vehicles have to sense all the TVWS channels before finding one available. In strategy B, the maximum delay is expected when the vehicles must remain listening to the channel while the fixed secondary network uses the TVWS channel in a sustained transmission equals to  $P_B$ . Reducing the access channel delay is a critical condition to meet the requirements of safety applications, and a desirable condition for more flexible applications.



Figure 4.7: Channel access delay for strategy A and strategy B depending on the probability of busy channel  $(P_B)$ 

The number of attempts before successful access to the TVWS channel using the two strategies affects the channel access delay. In strategy A, the number of attempts increases when  $P_B$  is higher because the vehicles have to move to another TVWS channel every time the vehicular network detects the fixed secondary network; in each attempt the vehicles must run the sensing mechanism which increase the channel access delay. In strategy B, an attempt is every time the vehicles remain listening to the TVWS channel until it becomes idle. Since strategy B does not execute a cooperative spectrum detection technique, there is no delay time caused by the spectrum detection and decision process.

Results in Fig. 4.8 show the packet loss rate. When  $P_B = 50\%$ , the maximum packet loss rate is 6% in strategy A and 5% in strategy B. When  $P_B = 70\%$ , the maximum packet loss rate is 9% in strategy A and 8% in strategy B. Hence, strategy B reduces the maximum packet loss rate by 17% in  $P_B=50\%$  and an 11% in  $P_B = 70\%$ , with respect to strategy A.



Figure 4.8: Packet loss rate for strategy A and strategy B depending on the probability of busy channel  $(P_b)$ 

#### 4.2.3 Discussion

According to the results, the vehicular network may share the TVWS channel with the fixed secondary network instead of searching for a new TVWS channel, as a strategy to reduce the channel access delay in opportunistic spectrum access. The reduction in the channel access delay impacts in the packet loss rate. For the average case of study, when the probability of busy channel is 50%, the strategy B reduces the maximum channel access delay by 15% and the maximum packet loss rate by 17%, regarding strategy A. For the near-to-worst case scenario, when the probability of busy channel is 70%, the strategy B reduces the maximum channel access the maximum channel access delay by 25% and the maximum packet loss rate by 11%, regarding strategy A. Reducing the channel access delay and the packet loss rate are critical requirements for many applications such as safety applications that tolerate a maximum delay (usually 100 ms), and beaconing application for which the perceived location precision is affected by the packet loss.

## 4.3 Performance Evaluation of the White Space Resource Sharing mechanism

#### 4.3.1 Evaluation scenario

The scenario employed to evaluate the proposed resource allocation mechanism's performance is shown in Fig. 4.9. A vehicular network travels across a route of 9 km where there are four non-overlapping FSNs deployed. The presence of FSNs (R) is defined as the percentage of active FSNs on the route using the same TVWS channel (e.g., R = 50% means that two FSNs are active on the route). The vehicular network has the same TVWS channel assigned as the FSNs to analyze the mechanism's performance depending on the active FSNs. In a situation in which vehicles travel across an area without an active FSN, the vehicular network uses the maximum transmit power, but in this scenario as vehicles enter the coverage area of an active FSN, the vehicular network applies the WSRS mechanism to share the TVWS channel.



Figure 4.9: Simulated scenario. Source: Own elaboration.

The evaluation considered two cases, shown in Fig. 4.10, for the distribution of the fixed nodes within the network coverage area: Case A is when the fixed nodes are far from the road uniformly distributed throughout the coverage area, whereas Case B is when the fixed nodes are near the road. Using the Montecarlo method, the effect of the fixed nodes' transmit power in the power control function and the feasibility of allowing fixed and mobile nodes to share the TVWS channel due to mutual interference is also analyzed. For this analysis, 21 simulations were done in Matlab per each fixed node distribution and fixed node transmit power. The simulation parameters are detailed in Table 4.7.

Table 4.7: Parameters for simulated scenario

Parameter	Value
N	10
M	10
f	710 MHz
$p_c^j$	$20~\mathrm{dBm}$ - $36~\mathrm{dBm}$
$SINR_V$	5 dB
d	25 m
$PL_0$	29.83 [110]
$\gamma$	2.04 [110]
$\sigma_S$	2.48 [110]
$W_j$	600 kHz
$t_r$	5 beacons per seconds
P <sub>max</sub>	20 dBm


Figure 4.10: Fixed node distribution within the network coverage area. Source: Own elaboration.

The proposed WSRS mechanism is compared with the scenario in which the vehicular network stops transmitting when the vehicles travel across an active FNS that uses the same TVWS channel as the vehicular network. Both scenarios are evaluated using the following metrics:

#### Cumulative Channel Capacity

This metric is the channel capacity achieved by the set of vehicles for each time frame, which is calculated as

$$X = \sum_{i=1}^{N-1} W log_2 \left( 1 + \frac{P}{I+N} \right),$$
 (4.6)

where the variables depend on whether the WSRS mechanism is used or not. Using the WSRS mechanism, a vehicle i requests the allocation of a time slot and the allowed transmit power for a time frame. Assuming that the allocated time slot is the same as the A-CPE j, the bandwidth W of the slot is  $W_j$ , the allowed transmit power P is  $p_i^j$  according to the power control function detailed in Section 3.4.2, the interference I caused by A-CPE j over the vehicle i is  $I_j^i$ , and the channel noise is  $N_i$ . In the scenario in which the WSRS mechanism is not used, vehicles only transmit in areas where there are no FSNs; then, the transmit power is the maximum (20 dBm), and there is no interference caused by an FSN.

#### Service Rate

A vehicle requests resource allocation in each time frame. The service rate is the ratio between the number of successful resource allocation requests  $(S_Q)$  and the total number of requests  $(T_Q)$ . It is calculated as

$$S_R = \frac{S_Q}{T_Q} \tag{4.7}$$

When the WSRS mechanism is not used and the vehicle travels within an active FSN coverage area, the service rate is zero; otherwise, the service rate is 1.

#### Jain's Index

Jain's Index is a fairness measure based on the offered channel capacity to the vehicles according to the allocated transmit power and channel access.

$$J(X) = \frac{\left(\sum_{i=1}^{N-1} X_i\right)^2}{(N-1)\sum_{i=1}^{N-1} X_i^2},$$
(4.8)

#### Spectral Efficiency

The spectral efficiency determines the spectrum utilization made by the entire system, which includes the vehicular network and the fixed secondary network. Spectral efficiency is the number of bits per second per hertz sent by all the channel users. For the vehicular network's TVWS channel utilization  $(X_V)$ , the channel capacity achieved due to the vehicular transmit power and the interference caused by the active FNSs when the WSRS mechanism is used; if the WSRS is not used, the vehicles use the maximum transmit power without additional interference. For the FNS's TVWS channel utilization  $(X_F)$ , the channel capacity depends on the A-CPEs transmit power and the interference caused by the vehicular network on the A-CPEs. The TVWS channel's bandwidth  $(W_{TVWS})$  is 6 MHz. Spectral efficiency is calculated as

$$E = \frac{X_V + X_F}{W_{TVWS}} \tag{4.9}$$

#### 4.3.2 Simulation Results

This section presents the evaluation performance of the White Space Resource Sharing (WSRS) mechanism in the simulated scenario (see Fig. 4.9) in Matlab, comparing the system using the WSRS mechanism and without the WSRS mechanism (non-WSRS). The non-WSRS scenario corresponds to the vehicular network conditions when the coexistence mechanism between the secondary fixed network and the vehicular network on the same TVWS channel is not enabled.



Figure 4.11: Normalized Average Channel Capacity per time frame for the set of vehicles

#### Normalized Cumulative Channel Capacity

The normalized cumulative instantaneous channel capacity for the set of 10 vehicles is shown in Fig 4.11a (Case A) and Fig 4.11b (Case B). The normalized cumulative instantaneous channel capacity is higher as the presence of secondary fixed networks (R) in the path decreases, regardless of the use of the WSRS mechanism or non-WSRS and the A-CPES distribution with respect to the road. When the WSRS mechanism is used, the vehicles transmit with maximum power (20 dBm) in areas with no fixed networks, then the cumulative channel capacity increases. In areas with fixed networks, the vehicles adjust their transmit power to reduce the interference caused to the fixed network. The cumulative channel capacity decreases due to the reduction in the vehicular transmit power and the interference received from the fixed secondary network.

The cumulative channel capacity decreases when the A-CPE transmit power is around 33 dBm for Case A (see Fig. 4.11a) and around 29 dBm for Case B (see Fig. 4.11b). In Case B, the A-CPEs are closer to the road, then the interference caused by the fixed network also increases, and more vehicular transmit power is required. If the vehicular transmit power is greater than the maximum power (20 dBm), the vehicle does not receive resource allocation, which affects the cumulated capacity for the set of vehicles. In the non-WSRS scenarios, the normalized average channel capacity is the same depending on the presence of the fixed secondary network. The A-CPE transmit power does not affect the channel capacity because the vehicles do not transmit in the areas where there is a fixed secondary network.

#### Service Rate

The service rate measures whether resources were allocated to a target vehicle or not during the time frame evaluated (see Fig. 4.12). The red line represents the service rate of the non-WSRS scenario depending on the presence of fixed secondary networks. As the presence of fixed secondary networks increases, the non-WSRS scenario's service rate decreases because the vehicles do not transmit in areas where there are fixed secondary networks. When the A-CPE transmit power increases, the distribution of the service rate varies. When the presence of the secondary fixed networks is 100% (R = 100%) and the A-CPE transmit power is 20 dBm, the service rate for the set of vehicles is 1.0, which means that all requests for resources from vehicles are serviced in both Case A (Fig. 4.12a) and Case B (Fig. 4.12b).

For Case A, when the A-CPE transmit power increases to 36 dBm (Fig. 4.12a-c), the service rate varies between 0.4 and 1.0, which indicates that not all requests for resource allocation from the vehicular network are met. For Case B, the service rate varies between 0.3 and 1.0 when the A-CPE transmit power is 36 dBm (Fig. 4.12b-c). Increasing the A-CPE transmit power also increases the interference caused by the fixed secondary network on the vehicular network. If the required vehicular transmit power is greater than the maximum power to face the interference of the fixed secondary network, the resource allocation for the vehicle cannot be met; therefore, the service rate decreases. Results show that the service rate is higher with the use of the WSRS mechanism, offering better conditions for the operation of the vehicular network, even in situations of coexistence and interference of the fixed secondary network shared use.



(a) Case A



(b) Case B

Figure 4.12: Service rate for the set of vehicles

#### Jain's Index

The A-CPE transmit power affects the Jain's Index of the system in both Case A and Case B, as shown in Fig. 4.13a and Fig. 4.13a respectively. As the transmit power increases, the interference caused by the fixed secondary network also increases, causing the vehicles to require a higher transmit power. If the required vehicular transmit power is higher than the maximum power, the vehicle will not receive the resource allocation. Therefore, the channel capacity offered to the vehicle is zero affecting the Jain's Index of the set of vehicles. For Case A, the A-CPEs are far from the road and the Jain's Index for the vehicular network is above 0.8. For Case B, where the A-CPEs are nearer to the road, the Jain's Index goes down to 0.6 because the interference of the fixed secondary network is higher than in Case A.

#### **Spectral Efficiency**

The spectral efficiency determines the spectrum utilization made by the entire system that includes the fixed secondary network and the vehicular network. The fixed network contributes the highest spectral efficiency in the system as the A-CPE transmit power increases. For both Case A (Fig. 4.14a) and Case B (Fig. 4.14b), the spectral efficiency of the system is higher as the presence of fixed secondary networks on the road increases. When the presence of secondary fixed networks is 100%, the spectral efficiency in the non-WSRS corresponds to the spectral efficiency achieved when the fixed secondary network is the only one using the TVWS channel.

The system's spectral efficiency increases when the sharing of the TVWS channel is allowed because the transmissions made by the vehicular network are added to the transmissions of the fixed secondary network. When the A-CPE transmit power is 20 dBm, the system's spectral efficiency is 17 bits/s/Hz. By increasing the A-CPE transmit power to 36 dBm, the system's spectral efficiency is 21 bits/s/Hz. Using the WSRS mechanism to share the TVWS channel, when R=100%, the spectral efficiency increases up to 32% (8 bits/s/Hz) with the transmissions of the vehicular network. The increment in the A-CPE transmit power also increases the spectral efficiency because this metric is related to the channel capacity (see Section 4.3.2 for more details of this metric). The effect of the interference caused by the vehicular network is taken into consideration in the spectral efficiency metric. Although the spectral efficiency achieved by the fixed secondary network decreases due to the interference caused by the vehicular network, the spectral efficiency provided by the vehicular network increases the spectral efficiency of the whole system.

Comparing Case A (A-CPEs furthest from the road) with Case B (A-CPEs closest to the road), the spectral efficiency curves follow the same trend. However, the spectral efficiency flattens out slightly at the end of the A-CPE transmit power range in Case B. Furthermore, in Case B, the channel capacity of the vehicular network decreases as the A-CPE transmit power increases (see Section 4.3.2 for more details of this metric), which decreases the contribution of the vehicular network to the spectral efficiency of the system.



Figure 4.13: Average Jain's Index per time frame for the set of vehicles



Figure 4.14: Spectral efficiency for the FNS and VN system

#### 4.3.3 Discussion

The WSRS mechanism was evaluated when vehicular networks move within an area where there are several fixed secondary networks, and there are not enough vacated TVWS channels to allocate, so overlapped fixed and vehicular networks must share the TVWS channel.

The offered channel capacity for the vehicular network is higher as the presence of active secondary fixed networks in the path decreases for both Case A and Case B, regardless of the use of the WSRS mechanism or non-WSRS. The channel capacity increases up to 40% using the WSRS mechanism when the presence of active fixed secondary networks along the road is 100%. The interference caused by the fixed secondary network affects the offered channel capacity for the vehicular network. Results showed the offered channel capacity begins to decrease when the A-CPE transmit power is around 33 dBm for Case A and around 29 dBm for Case B. If the vehicular transmit power required to face the fixed secondary network is greater than the maximum power (20 dBm), the vehicle does not receive resource allocation, which affects the channel capacity for the set of vehicles.

The service rate measures whether resources were allocated to a target vehicle or not during the time frame evaluated. As the presence of active fixed secondary networks increases, the non-WSRS scenario's service rate decreases because the vehicles do not transmit in areas where there are active fixed secondary networks. For Case A, when the A-CPE transmit power increases to 36 dBm, the service rate varies between 0.4 and 1.0, which indicates that not all requests for resource allocation from the vehicular network are met. For Case B, the service rate varies between 0.3 and 1.0 when the A-CPE transmit power is 36 dBm. Increasing the A-CPE transmit power also increases the interference caused by the fixed secondary network on the vehicular network. If the required vehicular transmit power is greater than the maximum power to face the interference of the fixed secondary network, the resource allocation for the vehicle cannot be met; therefore, the service rate decreases. The Jain's Index metric is also affected if vehicles do not receive the resource allocation when the required transmission power is higher than the maximum power. For Case A, the A-CPEs are far from the road and the Jain's Index for the vehicular network is above 0.8. For Case B, where the A-CPEs are nearer to the road, the Jain's Index goes down to 0.6 because the interference of the fixed secondary network is higher than in Case A.

The spectral efficiency determines the entire system's spectrum utilization, including the fixed secondary network and the vehicular network. The fixed secondary network contributes the highest spectral efficiency in the system as the A-CPE transmit power increases, because the channel capacity of the fixed secondary network increases, and active fixed secondary networks also increase. The system's spectral efficiency increases when the sharing of the TVWS channel is allowed because the transmissions made by the vehicular network are added to the transmissions of the fixed secondary network. When the A-CPE transmit power is 20 dBm, the system's spectral efficiency is 17 bits/s/Hz. By increasing the A-CPE transmit power to 36 dBm, the system's spectral efficiency is 21 bits/s/Hz. Using the WSRS mechanism to share the TVWS channel, when the 100% of the fixed secondary networks are active, the spectral efficiency increases up to 32% (8 bits/s/Hz) due to the transmissions of the vehicular network. Comparing Case A (A-CPEs furthest from the road) with Case B (A-CPEs closest to the road), the spectral efficiency curves follow the same trend. However, the spectral efficiency flattens out slightly at the end of the A-CPE transmit power range in

Case B. Furthermore, in Case B, the channel capacity of the vehicular network decreases as the A-CPE transmit power increases, as was explained before, which decreases the vehicular network's contribution to the system's spectral efficiency.

Results show that sharing the TVWS channel between fixed and mobile networks using the WSRS mechanism is feasible and increases the radio spectrum's efficiency. The WSRS mechanism is proper for scenarios where there are not enough vacated TVWS channels to allocate, so overlapped fixed and vehicular networks must share the TVWS channel. The WSRS mechanism's strengths are:

- The WSRS mechanism implements a Power Control Process to adjust the vehicles' transmission power and maintain vehicular communication, facing the interference caused by the fixed secondary network. The power control process respects the maximum power allowed for mobile devices in TVWS systems according to the spectrum regulation. This is a consideration also protects the primary TV systems in the event that vehicles move within the coverage area of TV systems.
- The WSRS mechanism considers the interference caused by the secondary fixed network, depending on the proximity of the A-CPEs to the vehicles. The Channel Allocation Process seeks to match each vehicular link with the farthest A-CPE as a strategy to reduce the mutual interference. The lower the interference caused by the secondary fixed network over the vehicular link, the lower the vehicle's transmission power. The decrease in the transmission power of the vehicles is a mechanism that protects the fixed secondary network from mutual interference.
- The WSRS mechanism is helpful for coexistence scenarios in other frequency bands in addition to the TVWS bands. The Power Control Process can be adjusted to include more detailed aspects of the channel model for both the mobile and fixed secondary network, but keeping the algorithm. The adjustments in the Power Control Process consist of changing the path loss models to calculate the variables  $L_{vi}$  (the path loss in the vehicular link i) and  $L_j^i$  (the path loss in the link between A-CPE j and vehicle i). The rest of the WSRS algorithm remains the same.
- The Channel Allocation Process is also flexible to implement different cost functions beyond maximizing the distance between the vehicle and A-CPE using the same subchannel to reduce the mutual interference. Another cost function to maximize could be the channel capacity in the vehicular network.

The WSRS mechanism's weaknesses are:

- The allocation of resources could constantly vary if the vehicular channel has a high path loss measure. In that case, the mechanism could not allocate resources if the required vehicular transmit power is higher than the maximum power, causing degradation in vehicular communication.
- Nor does the mechanism seek to guarantee the stable assignment of the same subchannel for vehicles. The WSRS mechanism does not consider the impact of the frequency shifting process on the TVWS radio interface.
- The WSRS mechanism was designed for scenarios where the number of vehicles and A-CPEs is the same. The mechanism does not require adjustments when the number of A-CPEs is greater than the number of vehicles, and the number of sub-channels is

equal to the number of A-CPEs. However, the WSRS mechanism has to be adjusted to support scenarios where the number of vehicles is greater than the number of subchannels. The adjustment could consist of dividing the sub-channel in the time-domain allowing more vehicles to use the same sub-channel as the A-CPE. The adjustment would require considering A-CPE interference on multiple vehicles in the Power Control Process. Also, it may require to adjust the cost function of the Channel Assignment Process to match multiple vehicles to one A-CPE.

## Chapter 5

## Conclusions

This chapter concludes the main research results and discuss the further work. The objectives were fulfilled with the design of the White Space Resource Sharing mechanism (WSRS). The WSRS mechanism allocates spectrum resources when not enough vacant TVWS channels are available and fixed secondary and vehicular networks must share a TVWS channel. Regarding to the specific objectives, a new metric was introduced, the Channel Availability for Opportunistic Vehicular Access (CAFOVA), which relates the channel occupancy of the fixed secondary network with random access channel, the speed of the vehicle, and the channel verification distance. This metric estimates the transmission opportunities for vehicular access when the vehicles move within the fixed secondary network's coverage area. The WSRS mechanism was evaluated using standard metrics such as offered channel capacity, service rate, Jain's Index, and spectral efficiency. Furthermore, the proposed mechanism was compared with the IEEE standard 802.22 mechanism.

The fulfillment of the hypotheses and the objectives are detailed below.

### 5.1 Major Research Results

The objective of this research is to investigate the coexistence among secondary networks for vehicular networking on the TVWS spectrum. The deployment of cognitive secondary networks over the TVWS spectrum creates a problem of coexistence because prior agreements for coordinating the sharing of channel usage are not required. Even though there are different methods and algorithms for coexistence between secondary cognitive networks, the complexity of the architecture and the non-obligatory adjustment of the operating parameters to achieve coexistence avoiding mutual interference have made it a challenge to enable the TVWS channel sharing. In what follows, a summary of the main research results are provided.

The first approach to explore the coexistence problem was to analyze the opportunistic use of the TVWS spectrum for vehicular communications when primary and IEEE 802.11af secondary fixed users are active. Primary and secondary TV band users were modeled, and several traffic profiles based on probability of busy channel were used to represent the TVWS channel occupation made by the fixed users.

A new metric called CAFOVA was introduce. The CAFOVA corresponds to the percentage of channel availability for opportunistic vehicular access when there are other opportunistic networks over the same TVWS channel. The CAFOVA relates the channel occupancy of the fixed secondary network with random access channel, the speed of the vehicle, and the channel verification distance. Results demonstrated that the CAFOVA depends on the TVWS channel occupation policy used by the IEEE 802.11af fixed secondary network (the White-Fi network). When the traffic in the White-Fi network follows a uniformly distributed probability of busy channel, the percentage of opportunities is around the complement of the probability of occupation. On the other hand, the CAFOVA was shown to be insensitive to bursty traffic occupation policies. This first approach demonstrated that vehicular networks may use the TVWS even if there are other opportunistic networks.

The vehicular network may share the TVWS channel with the fixed secondary network instead of searching for a new TVWS channel, as a strategy to reduce the channel access delay in opportunistic spectrum access. The reduction in the channel access delay impacts in the packet loss rate. For the average case of study, when the probability of busy channel is 50%, sharing the TVWS channel reduces the maximum channel access delay by 15% and the maximum packet loss rate by 17%, regarding searching for a new TVWS channel. For the near-to-worst case scenario, when the probability of busy channel is 70%, sharing the TVWS channel access delay by 25% and the maximum packet loss rate by 11%, regarding for a new TVWS channel. The reduction in the channel access delay and the packet loss rate are highlights to continue exploring the TVWS channel sharing as the strategy for vehicular communication on TVWS.

The second scenario of study was focused on the coexistence among overlapping IEEE 802.22 fixed and mobile secondary networks, when not enough vacant TVWS channels are available for allocation. Based on this scenario, the proposed White Space Resource Sharing mechanism (WSRS) was designed and modeled. The WSRS mechanism coordinates the channel access in the vehicular network according to the scheduling information of the fixed secondary network. The WSRS mechanism uses the flexibility of the cognitive vehicular network to improve network coexistence without requiring any changes in fixed network operating parameters. The WSRS mechanism implements a strategy to protect the fixed secondary network from interference caused by vehicular transmissions while promoting that both networks simultaneously use the channel.

The WSRS mechanism was evaluated when vehicular networks move within an area where there are several fixed secondary networks, and there are not enough vacated TVWS channels to allocate, so overlapped fixed and vehicular networks must share the TVWS channel. Simulation results show the offered channel capacity to the vehicular network increases up to 40% using the WSRS mechanism compared to offered channel capacity when TVWS channel sharing is not allowed, and no other TVWS channel is available. The WSRS mechanism allocates resources to vehicles even when the density of fixed nodes and the transmission power of the fixed network cause high interference to the vehicular network. Using the WSRS mechanism, the service rate varies between 0.3 and 1.0. When the TVWS channel sharing is not enabled, the service rate varies between zero and 0.8 because the vehicular network must stop transmitting when a fixed secondary network uses the TVWS channel. The system's spectral efficiency increases up to 32% when the vehicular network and the fixed secondary network overlap along the route, and both share the same TVWS channel.

### 5.2 Future Work

This research has investigated the coexistence among secondary networks for vehicular networking on the TVWS spectrum. Nevertheless, the are still several research directions in which this research can be extended.

- Estimation of the time that the TVWS channel is available at each channel access opportunity: The estimation of how much time the TVWS channel is available as a way to complement the CAFOVA metric. This estimate would be helpful to determine the channel capacity offered to the vehicular network at each channel access opportunity and to establish which additional services could use the TVWS channel.
- Coexistence among TV primary networks and vehicular secondary networks: The spectrum regulation defines that secondary networks cannot use the TV channels occupied by primary TV users to avoid harmful interference against the incumbent users. However, the proposed WSRS mechanism could be adapted to analyze the technical viability of facilitating TV channel sharing among primary and vehicular networks (or mobile groups in general). One alternative is to use the WSRS mechanism in areas where the vehicular network and the television network overlap, but the service level of the television network is below the sensitivity levels of the TV receivers. The WSRS mechanism could be adjusted to consider planning aspects of the TV network, such as the service contours of the transmitters.
- Supporting high-density fixed networks and variable transmit power: commercial initiatives and manufacturers are providing technological services for rural broadband connectivity, the industrial Internet of Things (IIoT), communications for critical infrastructure and environmental monitoring, among other applications, all operating on TVWS spectrum. These deployments have different characteristics, such as the number of nodes, density, mobility, and quality of service parameters. Because of the mobility of the vehicular network, the probability that the vehicular network and another secondary network overlap will increase. An alternative is to extend the strategies for resource allocation to support different secondary network profiles.
- Evaluation of the acceleration of the vehicles in the Power Control Process: In the proposed WSRS mechanism, the path loss model for the vehicular link could be changed to explicitly consider the acceleration parameter or other aspects of mobile channel propagation in the TVWS frequency band. This changes the calculation of the path loss in the vehicular link  $(L_{vi})$ , but the power control process in the Resource Allocation Function (RAF) remains the same.
- Present the proposed WSRS mechanism as a contribution to the IEEE standard 802.22 Working Group: The IEEE 802.22 standard indicates only that when

no vacated TWVS channels are available for allocation, or when different groups share the TVWS channel for spectrum efficiency purposes, then "the A-BS may use a mechanism for proper resource sharing between the overlapping groups or the A-BS and some groups after channel switching" [23]. However, such a resource sharing mechanism is neither detailed nor defined by the standard. The proposed WSRS mechanism may be a baseline to evaluate different use cases and adjust the power control or the channel access strategy according to the expected standard evolution.

# Chapter 6

# Appendix

Main code repository: WSRS mechanism's code

## Bibliography

- [1] National Highway Traffic Safety Administration, "Federal motor vehicle safety standards; v2miorandiv communications," *Proposed Rules*, vol. 82, no. 8, 2017.
- [2] Markets and Markets, "Connected Car Market by Service (OTA Update, Navigation, Cybersecurity, Multimedia Streaming, Social Media, e-Call, Autopilot, Home Integration & Other), Form, End Market (OE, Aftermarket), Network, Transponder, Hardware and Region - Global Forecast to 2025," Markets and Markets, 2020.
- [3] Business Insider Intelligence, "How 5g & iot technologies are driving the connected smart vehicle industry," *Business Insider Intelligence*, 2020.
- [4] Intel Newsroom, "Data is the new oil in the future of automated driving," Intel Newsroom, 2016.
- [5] G. Karagiannis, O. Altintas, E. Ekici, G. Heijenk, B. Jarupan, K. Lin, and T. Weil, "Vehicular networking: A survey and tutorial on requirements, architectures, challenges, standards and solutions," *IEEE Communications Surveys Tutorials*, vol. 13, pp. 584– 616, Fourth 2011.
- [6] J. B. Kenney, "Dedicated short-range communications (DSRC) standards in the united states," *Proceedings of the IEEE*, vol. 99, pp. 1162–1182, July 2011.
- [7] X. Wang, S. Mao, and M. X. Gong, "An Overview of 3GPP Cellular Vehicle-to-Everything Standards," *GetMobile: Mobile Comp. and Comm.*, vol. 21, p. 19–25, Nov. 2017.
- [8] FCC News for the Federal Communications Commission, "FCC modernizes 5.9 GHz band for Wi-Fi and auto safety," ET Docket No. 19-138, U.S. Federal Communications Commission, Nov. 2020.
- [9] European Telecommunications Standards Institute, "Road transport and traffic telematics - dedicated short range communication (DSRC) - DSRC data link layer: Medium access and logical link control," standard, ETSI, Mar. 2003.
- [10] Association of Radio Industries and Businesses, "Dedicated short range communication (DSRC)," standard, ARIB, Sept. 2001.
- [11] D. Evans, "Spectrum scarcity? what spectrum scarcity?," Sep 2020.

- [12] Y. Han, E. Ekici, H. Kremo, and O. Altintas, "A survey of mac issues for tv white space access," Ad Hoc Networks, vol. 27, pp. 195 – 218, 2015.
- [13] Q. Zhao and B. M. Sadler, "A survey of dynamic spectrum access," *IEEE Signal Processing Magazine*, vol. 24, pp. 79–89, May 2007.
- [14] A. Arteaga, S. Céspedes, and C. Azurdia-Meza, "Vehicular Communications Over TV White Spaces in the Presence of Secondary Users," *IEEE Access*, vol. 7, pp. 53496– 53508, 2019.
- [15] C. Mikeka, J. S. Mlatho, M. Thodi, J. Pinifolo, D. Kondwani, L. Momba, M. Zennaro, A. Arcia-Moret, C. Fonda, and E. Pietrosemoli, "Preliminary performance assessment of TV white spaces technology for broadband communication in malawi," *Proceedia Engineering*, vol. 78, pp. 149 – 154, 2014.
- [16] M. Shapiro, D. Murphy, A. Yankelevich, and W. H. Dutton, "Wireless innovation for last mile access: An analysis of cases and business strategies," Dec. 2016.
- [17] A. A. Lysko, M. T. Masonta, M. R. O. Mofolo, L. Mfupe, L. Montsi, D. L. Johnson, F. Mekuria, D. W. Ngwenya, N. S. Ntlatlapa, A. Hart, C. Harding, and A. Lee, "First large TV white spaces trial in south africa: A brief overview," in 2014 6th International Congress on Ultra Modern Telecommunications and Control Systems and Workshops (ICUMT), pp. 407–414, Oct 2014.
- [18] O. Holland, N. Sastry, S. Ping, R. Knopp, F. Kaltenberger, D. Nussbaum, J. Hallio, M. Jakobsson, J. Auranen, R. Ekman, J. Paavola, A. Kivinen, H. N. Tran, K. Ishizu, H. Harada, P. Chawdhry, J. M. Chareau, J. Bishop, M. Bavaro, E. Anguili, Y. Gao, R. Dionisio, P. Marques, H. Kokkinen, and O. Luukkonen, "A series of trials in the UK as part of the Ofcom TV white spaces pilot," in 2014 1st International Workshop on Cognitive Cellular Systems (CCS), pp. 1–5, Sept 2014.
- [19] A. Montejo, A. Bañacia, H. Sawada, K. Ishizu, K. Ibuka, and F. Kojima, "Implementation of a multi-hop network at the university campus using an IEEE 802.11afcompliant network," in 2017 20th International Symposium on Wireless Personal Multimedia Communications (WPMC), pp. 173–180, Dec 2017.
- [20] O. Altintas, K. Seki, H. Kremo, M. Matsumoto, R. Onishi, and H. Tanaka, "Vehicles as information hubs during disasters: Glueing Wi-Fi to TV white space to cellular networks," *IEEE Intelligent Transportation Systems Magazine*, vol. 6, pp. 68–71, Spring 2014.
- [21] D. Vasisht, Z. Kapetanovic, J.-h. Won, X. Jin, R. Chandra, A. Kapoor, S. N. Sinha, M. Sudarshan, and S. Stratman, "Farmbeats: An IoT platform for data-driven agriculture," in *Proceedings of the 14th USENIX Conference on Networked Systems Design* and Implementation, NSDI'17, (Berkeley, CA, USA), pp. 515–528, USENIX Association, 2017.
- [22] IEEE, "IEEE Standard for Information technology–Telecommunications and infor-

mation exchange between systems–Local and metropolitan area networks–Specific requirements–Part 19: Wireless Network Coexistence Methods," *IEEE Std 802.19.1-2018*, pp. 1–458, 2018.

- [23] IEEE, "IEEE Standard Information Technology-Telecommunications and information exchange between systems-Wireless Regional Area Networks-Specific requirements-Part 22: Cognitive Wireless RAN MAC and PHY specifications: Policies and Procedures for Operation in the Bands that Allow Spectrum Sharing where the Communications Devices May Opportunistically Operate in the Spectrum of Primary Service," *IEEE Std 802.22-2019 (Revision of IEEE Std 802.22-2011)*, pp. 1–1465, 2020.
- [24] World Health Organization, "Global Status Report on Road Safety 2015," Report, 2015.
- [25] Z. Cheng, M.-S. Pang, and P. Pavlou, "Mitigating traffic congestion: The role of intelligent transportation systems," *Information Systems Research*, vol. 31, no. 3, pp. 653– 674, 2020.
- [26] K. Kusano and H. Gabler, "Comparison of expected crash and injury reduction from production forward collision and lane departure warning systems," *Traffic injury prevention*, vol. 16, no. sup. 2, pp. S109–S114, 2016.
- [27] Y. Toor, P. Muhlethaler, A. Laouiti, and A. D. L. Fortelle, "Vehicle ad hoc networks: applications and related technical issues," *IEEE Communications Surveys Tutorials*, vol. 10, pp. 74–88, Third 2008.
- [28] K. Zheng, Q. Zheng, P. Chatzimisios, W. Xiang, and Y. Zhou, "Heterogeneous vehicular networking: A survey on architecture, challenges, and solutions," *IEEE Communications Surveys Tutorials*, vol. 17, pp. 2377–2396, Fourthquarter 2015.
- [29] IEEE, "IEEE standard for information technology-telecommunications and information exchange between systems local and metropolitan area networks-specific requirements part 11: Wireless LAN medium access control (MAC) and physical layer (PHY) specifications," *IEEE Std 802.11-2012 (Revision of IEEE Std 802.11-2007)*, pp. 1–2793, March 2012.
- [30] 3GPP, "Evolved Universal Terrestrial Radio Access (E-UTRA) and Evolved Universal Terrestrial Radio Access Network (E-UTRAN);," overall description; stage 2 (v14.3.0, release 14), 3GPP, June 2017.
- [31] H. Seo, K. Lee, S. Yasukawa, Y. Peng, and P. Sartori, "LTE evolution for vehicle-toeverything services," *IEEE Communications Magazine*, vol. 54, pp. 22–28, June 2016.
- [32] 3GPP, "Study on enhancement of 3GPP support for 5G V2X services (v15.0.0, release 15)," tech. rep. 22.886, 3GPP, Mar. 2017.
- [33] R. Molina-Masegosa and J. Gozalvez, "LTE-V for sidelink 5G V2X vehicular communications: A new 5G technology for short-range vehicle-to-everything communications," *IEEE Vehicular Technology Magazine*, vol. 12, pp. 30–39, Dec 2017.

- [34] N. Cheng, N. Lu, N. Zhang, X. Shen, and J. W. Mark, "Vehicular WiFi offloading: Challenges and solutions," *Vehicular Communications*, vol. 1, pp. 13–21, 2014.
- [35] N. Lu, N. Cheng, N. Zhang, X. S. Shen, J. W. Mark, and F. Bai, "Wi-fi hotspot at signalized intersection: Cost-effectiveness for vehicular internet access," *IEEE Transactions on Vehicular Technology*, vol. 65, pp. 3506–3518, May 2016.
- [36] B. Coll-Perales, M. Gruteser, and J. Gozalvez, "Evaluation of IEEE 802.11ad for mmWave V2V communications," in 2018 IEEE Wireless Communications and Networking Conference Workshops (WCNCW), pp. 290–295, April 2018.
- [37] E. Grossi, M. Lops, L. Venturino, and A. Zappone, "Opportunistic radar in ieee 802.11ad vehicular networks," in 2017 IEEE 85th Vehicular Technology Conference (VTC Spring), pp. 1–5, June 2017.
- [38] I. Mavromatis, A. Tassi, R. J. Piechocki, and A. Nix, "mmWave System for Future ITS: A MAC-Layer Approach for V2X Beam Steering," in 2017 IEEE 86th Vehicular Technology Conference (VTC-Fall), pp. 1–6, Sept 2017.
- [39] IEEE, "ISO/IEC/IEEE International Standard for Information technology– Telecommunications and information exchange between systems–Local and metropolitan area networks–Specific requirements-Part 11: Wireless LAN Medium Access Control (MAC) and Physical Layer (PHY) Specifications Amendment 3: Enhancements for Very High Throughput in the 60 GHz Band (adoption of IEEE Std 802.11ad-2012)," ISO/IEC/IEEE 8802-11:2012/Amd.3:2014(E), pp. 1–634, March 2014.
- [40] H. Haas, L. Yin, Y. Wang, and C. Chen, "What is lifi?," Journal of Lightwave Technology, vol. 34, pp. 1533–1544, March 2016.
- [41] N. A. Abdulsalam, R. A. Hajri, Z. A. Abri, Z. A. Lawati, and M. M. Bait-Suwailam, "Design and implementation of a vehicle to vehicle communication system using lifi technology," in 2015 International Conference on Information and Communication Technology Research (ICTRC), pp. 136–139, May 2015.
- [42] I. Takai, T. Harada, M. Andoh, K. Yasutomi, K. Kagawa, and S. Kawahito, "Optical vehicle-to-vehicle communication system using led transmitter and camera receiver," *IEEE Photonics Journal*, vol. 6, pp. 1–14, Oct 2014.
- [43] J. Liu, P. Chan, D. Ng, E. Lo, and S. Shimamoto, "Hybrid visible light communications in intelligent transportation systems with position based services," in 2012 IEEE Globecom Workshops, pp. 1254–1259, Dec 2012.
- [44] J. Chen, B. Liu, H. Zhou, L. Gui, N. Liu, and Y. Wu, "Providing vehicular infotainment service using VHF/UHF TV bands via spatial spectrum reuse," *TBC*, vol. 61, no. 2, pp. 279–289, 2015.
- [45] O. Altintas, T. Watanabe, H. Kremo, H. Tanaka, H. Nakao, K. Tsukamoto, and M. Tsuru, "Design and experimental evaluation of a database-assisted V2V commu-

nications system over TV white space," J. Signal Process. Syst., vol. 83, pp. 45–55, Apr. 2016.

- [46] H. Zhou, N. Cheng, Q. Yu, X. S. Shen, D. Shan, and F. Bai, "Toward multi-radio vehicular data piping for dynamic DSRC/TVWS spectrum sharing," *IEEE Journal on Selected Areas in Communications*, vol. 34, pp. 2575–2588, Oct 2016.
- [47] H. Kremo, O. Altintas, H. Tanaka, M. Kitamura, K. Inage, and T. Fujii, "Cooperative spectrum sensing in the vehicular environment: An experimental evaluation," in 2014 IEEE 79th Vehicular Technology Conference (VTC Spring), pp. 1–5, May 2014.
- [48] X. Qian and L. Hao, "Spectrum sensing with energy detection in cognitive vehicular ad hoc networks," in 2014 IEEE 6th International Symposium on Wireless Vehicular Communications (WiVeC 2014), pp. 1–5, Sept 2014.
- [49] Y. Han, E. Ekici, H. Kremo, and O. Altintas, "Resource allocation algorithms supporting coexistence of cognitive vehicular and IEEE 802.22 networks," *IEEE Transactions* on Wireless Communications, vol. 16, pp. 1066–1079, Feb 2017.
- [50] H. Zhou, N. Cheng, N. Zhang, S. Wu, L. Gui, Q. Yu, X. S. Shen, and F. Bai, "Enabling efficient and wide-coverage vehicular content distribution over TV white spaces," in 2015 International Conference on Wireless Communications Signal Processing (WCSP), pp. 1–6, Oct 2015.
- [51] Federal Communications Commission, "Amendment of the commission's rules regarding dedicated short-range communication services in the 5.850–5.925 GHz band (5.9 GHz band)," FCC 06-110, U.S. Federal Communications Commission MO&O, July 2016.
- [52] M. D. Felice, R. Doost-Mohammady, K. R. Chowdhury, and L. Bononi, "Smart radios for smart vehicles: Cognitive vehicular networks," *IEEE Vehicular Technology Magazine*, vol. 7, pp. 26–33, June 2012.
- [53] K. D. Singh, P. Rawat, and J.-M. Bonnin, "Cognitive Radio for Vehicular AD-HOC Networks (CR-VANETs): Approaches and Challenges," *EURASIP Journal on Wireless Communications and Networking*, vol. 2014, p. 49, Mar 2014.
- [54] J. Mitola and G. Q. Maguire, "Cognitive radio: making software radios more personal," *IEEE Personal Communications*, vol. 6, pp. 13–18, Aug 1999.
- [55] D. F. Macedo, D. Guedes, L. F. M. Vieira, M. A. M. Vieira, and M. Nogueira, "Programmable networks – From software-defined radio to software-defined networking," *IEEE Communications Surveys Tutorials*, vol. 17, pp. 1102–1125, Secondquarter 2015.
- [56] I. Ku, Y. Lu, M. Gerla, R. L. Gomes, F. Ongaro, and E. Cerqueira, "Towards softwaredefined VANET: Architecture and services," in 2014 13th Annual Mediterranean Ad Hoc Networking Workshop (MED-HOC-NET), pp. 103–110, June 2014.
- [57] C. Jiacheng, Z. Haibo, Z. Ning, Y. Peng, G. Lin, and S. Xuemin (Sherman), "Software defined internet of vehicles: architecture, challenges and solutions," *Journal of*

Communications and Information Networks, vol. 1, pp. 14–26, Jun 2016.

- [58] Z. He, J. Cao, and X. Liu, "SDVN: enabling rapid network innovation for heterogeneous vehicular communication," *IEEE Network*, vol. 30, pp. 10–15, July 2016.
- [59] I. Yaqoob, I. Ahmad, E. Ahmed, A. Gani, M. Imran, and N. Guizani, "Overcoming the key challenges to establishing vehicular communication: Is SDN the answer?," *IEEE Communications Magazine*, vol. 55, no. 7, pp. 128–134, 2017.
- [60] W. Quan, K. Wang, Y. Liu, N. Cheng, H. Zhang, and X. S. Shen, "Software-Defined Collaborative Offloading for Heterogeneous Vehicular Networks," *Wireless Communi*cations and Mobile Computing, vol. 2018, 2018.
- [61] J. Chen, H. Zhou, N. Zhang, W. Xu, Q. Yu, L. Gui, and X. Shen, "Service-oriented dynamic connection management for software-defined internet of vehicles," *IEEE Transactions on Intelligent Transportation Systems*, vol. 18, pp. 2826–2837, Oct 2017.
- [62] K. Zheng, L. Hou, H. Meng, Q. Zheng, N. Lu, and L. Lei, "Soft-defined heterogeneous vehicular network: architecture and challenges," *IEEE Network*, vol. 30, pp. 72–80, July 2016.
- [63] Q. Zheng, K. Zheng, H. Zhang, and V. C. M. Leung, "Delay-optimal virtualized radio resource scheduling in software-defined vehicular networks via stochastic learning," *IEEE Transactions on Vehicular Technology*, vol. 65, pp. 7857–7867, Oct 2016.
- [64] K. Zheng, L. Hou, H. Meng, Q. Zheng, N. Lu, and L. Lei, "Soft-defined heterogeneous vehicular network: architecture and challenges," *IEEE Network*, vol. 30, pp. 72–80, July 2016.
- [65] J. Gozalvez, M. Sepulcre, and R. Bauza, "IEEE 802.11p vehicle to infrastructure communications in urban environments," *IEEE Communications Magazine*, vol. 50, pp. 176–183, May 2012.
- [66] K. A. Hafeez, L. Zhao, B. Ma, and J. W. Mark, "Performance Analysis and Enhancement of the DSRC for VANETs Safety Applications," *IEEE Transactions on Vehicular Technology*, vol. 62, pp. 3069–3083, Sept 2013.
- [67] X. Ma, X. Chen, and H. H. Refai, "Performance and Reliability of DSRC Vehicular Safety Communication: A Formal Analysis," *EURASIP Journal on Wireless Communications and Networking*, vol. 2009, p. 969164, Jan 2009.
- [68] Y. Han, Y. Wen, W. Tang, and S. Li, "Spectrum occupancy measurement: Focus on the TV frequency," in 2010 2nd International Conference on Signal Processing Systems, vol. 2, pp. 490–494, July 2010.
- [69] M. T. Masonta, M. Mzyece, and N. Ntlatlapa, "Spectrum Decision in Cognitive Radio Networks: A Survey," *IEEE Communications Surveys & Tutorials*, vol. 15, no. 3, pp. 1088–1107, 2013.

- [70] Federal Communications Commission, "Second report and order and memorandum opinion and order in ET docket nos. 02-380 (additional spectrum for unlicensed devices below 900 MHz and in the 3 GHz band) and 04-186 (unlicensed operation in the TV broadcast bands)," tech. rep., Federal Communications Commission, 2008.
- [71] Conference Europeenne des Administrations des Postes et des Télécommunications, "ECC report 159: Technical and operational requirements for the possible operation of cognitive radio systems in the white spaces of the frequency band 470-790 MHz," tech. rep., Conference Europeenne des Administrations des Postes et des Télécommunications, 2011.
- [72] S. Chen, Vehicular Dynamic Spectrum Access: Using Cognitive Radio for Automobile Networks. PhD thesis, Worcester Polytechnic Institute, 2012.
- [73] Federal Communications Commission, "White space database administrators guide."
- [74] H. Zhou, N. Cheng, N. Lu, L. Gui, D. Zhang, Q. Yu, F. Bai, and X. S. Shen, "Whitefi infostation: Engineering vehicular media streaming with geolocation database," *IEEE Journal on Selected Areas in Communications*, vol. 34, pp. 2260–2274, Aug 2016.
- [75] T. Yucek and H. Arslan, "A survey of spectrum sensing algorithms for cognitive radio applications," *IEEE Communications Surveys Tutorials*, vol. 11, pp. 116–130, First 2009.
- [76] M. A. Adulsattar and Z. A. Hussein, "Energy detection technique for spectrum sensing in cognitive radio: A survey," ussein, "Energy Detection Technique for Spectrum Sensing in Cognitive Radio: A survey", International Journal of Computer Networks & Communications (IJCNC), vol. 4, pp. 224–242, Sept. 2012.
- [77] H. Li and D. K. Irick, "Collaborative spectrum sensing in cognitive radio vehicular ad hoc networks: Belief propagation on highway," in 2010 IEEE 71st Vehicular Technology Conference, pp. 1–5, May 2010.
- [78] M. D. Felice, K. R. Chowdhury, and L. Bononi, "Cooperative spectrum management in cognitive vehicular ad hoc networks," in 2011 IEEE Vehicular Networking Conference (VNC), pp. 47–54, Nov 2011.
- [79] X. Qian and L. Hao, "On the performance of spectrum sensing in cognitive vehicular networks," in 2015 IEEE 26th Annual International Symposium on Personal, Indoor, and Mobile Radio Communications (PIMRC), pp. 1002–1006, Aug 2015.
- [80] S. Pagadarai, A. M. Wyglinski, and R. Vuyyuru, "Characterization of vacant UHF TV channels for vehicular dynamic spectrum access," in 2009 IEEE Vehicular Networking Conference (VNC), pp. 1–8, Oct 2009.
- [81] O. Altintas, M. Nishibori, T. Oshida, C. Yoshimura, Y. Fujii, K. Nishida, Y. Ihara, M. Saito, K. Tsukamoto, M. Tsuru, Y. Oie, R. Vuyyuru, A. A. Abbasi, M. Ohtake, M. Ohta, T. Fujii, S. Chen, S. Pagadarai, and A. M. Wyglinski, "Demonstration of vehicle to vehicle communications over TV white space," in 2011 IEEE Vehicular Tech-

nology Conference (VTC Fall), pp. 1–3, Sept 2011.

- [82] Y. Ihara, H. Kremo, O. Altintas, H. Tanaka, M. Ohtake, T. Fujii, C. Yoshimura, K. Ando, K. Tsukamoto, M. Tsuru, and Y. Oie, "Distributed autonomous multi-hop vehicle-to-vehicle communications over TV white space," in 2013 IEEE 10th Consumer Communications and Networking Conference (CCNC), pp. 336–344, Jan 2013.
- [83] M. Oner and F. Jondral, "Cyclostationarity based air interface recognition for software radio systems," in *Proceedings. 2004 IEEE Radio and Wireless Conference (IEEE Cat.* No.04TH8746), pp. 263–266, Sept 2004.
- [84] K. Kim, I. A. Akbar, K. K. Bae, J.-S. Um, C. M. Spooner, and J. H. Reed, "Cyclostationary approaches to signal detection and classification in cognitive radio," in 2007 2nd IEEE International Symposium on New Frontiers in Dynamic Spectrum Access Networks, Institute of Electrical and Electronics Engineers (IEEE), apr 2007.
- [85] P. D. Sutton, J. Lotze, K. E. Nolan, and L. E. Doyle, "Cyclostationary signature detection in multipath rayleigh fading environments," in 2007 2nd International Conference on Cognitive Radio Oriented Wireless Networks and Communications, pp. 408–413, Aug 2007.
- [86] Y. Zhu, K. Su, D. Wang, M. Zhu, and Y. Li, "Analysis of improved cyclostationary detector with SLC diversity over nakagami-m fading channels," in 2013 8th International Conference on Communications and Networking in China (CHINACOM), pp. 133–137, Aug 2013.
- [87] D. Romero, R. López-Valcarce, and G. Leus, "Generalized matched filter detector for fast fading channels," in 2012 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pp. 3165–3168, March 2012.
- [88] M. Matinmikko, T. Rauma, M. Mustonen, and J. D. Ser, "Architecture and approach for obtaining spectrum availability information," in 2011 IEEE 73rd Vehicular Technology Conference (VTC Spring), pp. 1–5, May 2011.
- [89] R. Almesaeed, N. F. Abdullah, A. Doufexi, and A. R. Nix, "A throughput study of white-fi networks in rural environment under realistic conditions and mobility," in 2015 IEEE 81st Vehicular Technology Conference (VTC Spring), pp. 1–5, May 2015.
- [90] H. Rasheed and N. Rajatheva, "Spectrum sensing for cognitive vehicular networks over composite fading," *International Journal of Vehicular Technology*, vol. 2011, pp. 1–9, 2011.
- [91] S. Atapattu, C. Tellambura, H. Jiang, and N. Rajatheva, "Unified analysis of low-SNR energy detection and threshold selection," *IEEE Transactions on Vehicular Technology*, vol. 64, pp. 5006–5019, nov 2015.
- [92] K. Tsukamoto, Y. Oie, H. Kremo, O. Altintas, H. Tanaka, and T. Fujii, "Implementation and performance evaluation of distributed autonomous multi-hop vehicle-to-vehicle communications over TV white space," *Mobile Networks and Applications*, vol. 20,

pp. 203–219, feb 2015.

- [93] J. H. Lim, W. Kim, K. Naito, and M. Gerla, "Interplay between TVWS and DSRC: Optimal strategy for QoS of safety message dissemination in VANET," in 2013 International Conference on Computing, Networking and Communications (ICNC), pp. 1156– 1161, Jan 2013.
- [94] S. Montejo-Sánchez, C. Azurdia-Meza, S. Bolufé, S. Céspedes, I. Soto, and R. D. Souza, "Novel channel hopping sequence approaches to rendezvous for vanets," in 2017 CHILEAN Conference on Electrical, Electronics Engineering, Information and Communication Technologies (CHILECON), pp. 1–5, Oct 2017.
- [95] D. Niyato, E. Hossain, and P. Wang, "Optimal channel access management with qos support for cognitive vehicular networks," *IEEE Transactions on Mobile Computing*, vol. 10, pp. 573–591, April 2011.
- [96] H. Zhou, N. Zhang, Y. Bi, Q. Yu, X. S. Shen, D. Shan, and F. Bai, "Tv white space enabled connected vehicle networks: Challenges and solutions," *IEEE Network*, vol. 31, pp. 6–13, May 2017.
- [97] A. Bourdena, E. Pallis, G. Kormentzas, and G. Mastorakis, "Efficient radio resource management algorithms in opportunistic cognitive radio networks," *Transactions on Emerging Telecommunications Technologies*, vol. 25, pp. 785–797, August 2014.
- [98] R. Ouyang, T. Matsumura, K. Mizutani, and H. Harada, "A reliable channel estimation scheme using scattered pilot pattern for ieee 802.22-based mobile communication system," *IEEE Transactions on Cognitive Communications and Networking*, vol. 5, no. 4, pp. 935–948, 2019.
- [99] IEEE, "IEEE std 802.11af-2013, Part 11: Wireless LAN medium access control (MAC) and physical layer (PHY) specifications amendment 5: Television white spaces (TVWS) operation," standard, IEEE, Feb. 2014.
- [100] IEEE, "IEEE std 802.22-2011, IEEE standard for information technology- local and metropolitan area networks- specific requirements- part 22: Cognitive wireless RAN medium access control (MAC) and physical layer (PHY) specifications: Policies and procedures for operation in the TV bands," standard, IEEE, July 2011.
- [101] S. Filin, T. Baykas, H. Harada, F. Kojima, and H. Yano, "IEEE standard 802.19.1 for TV white space coexistence," *IEEE Communications Magazine*, vol. 54, pp. 22–26, March 2016.
- [102] European Telecommunications Standards Institute, "ETSI, White Space Devices (WSD); Wireless Access Systems operating in the 470 MHz to 790 MHz TV broadcast band; Harmonized EN covering the essential requirements of article 3.2 of the R&TTE directive," techical report, ETSI, Apr. 2014.
- [103] H. Zhou, B. Liu, T. H. Luan, F. Hou, L. Gui, Y. Li, Q. Yu, and X. Shen, "Chaincluster: Engineering a cooperative content distribution framework for highway vehicular

communications," *IEEE Transactions on Intelligent Transportation Systems*, vol. 15, pp. 2644–2657, Dec 2014.

- [104] A. Arteaga, S. Céspedes, and C. Azurdia-Meza, "Next generation vehicular communications via interworking of DSRC and TV white space," in 2018 18th International Symposium on Antenna Technology and Applied Electromagnetics (ANTEM), pp. 1–2, Aug 2018.
- [105] N. Faruk, A. Ayeni, and Y. A. Adediran, "On the study of empirical path loss models for accurate prediction of TV signal for secondary users," *Progress In Electromagnetics Research B*, vol. 49, pp. 155–176, 2013.
- [106] M. Hata, "Empirical formula for propagation loss in land mobile radio services," IEEE Transactions on Vehicular Technology, vol. 29, no. 3, pp. 317–325, 1980.
- [107] P. Pathania, P. Kumar, and S. Rana, "Performance evaluation of different path loss models for broadcasting applications," *American Journal of Engineering Research* (AJER), vol. 3, pp. 335–342, Apr 2014.
- [108] D. Miorandi, A. A. Kherani, and E. Altman, "A queueing model for HTTP traffic over IEEE 802.11 WLANs," *Comput. Netw.*, vol. 50, pp. 63–79, Jan. 2006.
- [109] P. Rajesh, N. Kshirasagar, and S. Ajit, "Anatomy of WiFi access traffic of smartphones and implications for energy saving techniques," *International Journal of Energy, Information and Communications*, vol. 3, pp. 1–16, Feb. 2012.
- [110] H. Fernández, L. Rubio, V. M. Rodrigo-Peñarrocha, and J. Reig, "Path loss characterization for vehicular communications at 700 MHz and 5.9 GHz under LOS and NLOS conditions," *IEEE Antennas and Wireless Propagation Letters*, vol. 13, pp. 931–934, 2014.
- [111] D. Bertsekas, Linear Network Optimization: Algorithms and Codes. Cambridge, MA, USA: MIT Press, 1991.
- [112] ETSI, "Access, terminals, transmission and multiplexing (ATTM); system characteristics of receiver equipment installed in headends of integrated broadband cable and television networks intended to receive broadcast signals in the frequency range 470 MHz to 790 MHz," technical specification, ETSI, Nov. 2011.
- [113] A. Abuashour and M. Kadoch, "Performance improvement of cluster-based routing protocol in vanet," *IEEE Access*, vol. 5, pp. 15354–15371, 2017.
- [114] G. Yang, J. Wang, J. Luo, O. Y. Wen, H. Li, Q. Li, and S. Li, "Cooperative spectrum sensing in heterogeneous cognitive radio networks based on normalized energy detection," *IEEE Transactions on Vehicular Technology*, vol. 65, no. 3, pp. 1452–1463, 2016.
- [115] E. O. Guerra, V. A. Reguera, R. D. Souza, E. G. Fernández, and M. E. Pellenz, "Systematic construction of common channel hopping rendezvous strategies in cognitive radio

networks," *EURASIP Journal on Wireless Communications and Networking*, vol. 2015, p. 134, May 2015.

- [116] P. Palacios, A. Castro, C. Azurdia-Meza, and C. Estevez, "SVD detection analysis in cognitive mobile radio networks," in 2017 Ninth International Conference on Ubiquitous and Future Networks (ICUFN), pp. 222–224, July 2017.
- [117] A. Arteaga, P. Palacios, S. Céspedes, and C. A. Azurdia-Meza, "Evaluation of Opportunistic Access Strategies for Vehicular Networking Over TVWS," in 2020 IEEE 92nd Vehicular Technology Conference (VTC2020-Fall), pp. 1–5, 2020.
- [118] X. Ma, X. Chen, and H. H. Refai, "Performance and Reliability of DSRC Vehicular Safety Communication: A Formal Analysis," *EURASIP Journal on Wireless Communications and Networking*, vol. 2009, p. 969164, Jan 2009.
- [119] P. Palacios and C. Saavedra, "Coalition game theory in cognitive mobile radio networks," in *Technology Trends* (M. Botto-Tobar, G. Pizarro, M. Zúñiga-Prieto, M. D'Armas, and M. Zúñiga Sánchez, eds.), (Cham), pp. 3–15, Springer International Publishing, 2019.